

FISHERIES FINAL REPORT SERIES | NO. 145

Changes in fish and crustacean assemblages in tidal creeks of Hexham Swamp following the staged opening of **Ironbark Creek floodgates**

Craig Boys





www.dpi.nsw.gov.au

Published by the NSW Department of Primary Industries

Changes in fish and crustacean assemblages in tidal creeks of Hexham Swamp following the staged opening of Ironbark Creek floodgates, NSW DPI – Fisheries Final Report Series No. 145

First published June 2015

ISSN 2204-8669*

More information

Dr Craig Boys, Port Stephens Fisheries Institute, Locked Bag 1, Nelson Bay, NSW 2315, Australia

www.dpi.nsw.gov.au

Acknowledgments



Cover image: NSW Department of Primary Industries

© State of New South Wales through the Department of Trade and Investment, Regional Infrastructure and Services, 2015. You may copy, distribute and otherwise freely deal with this publication for any purpose, provided that you attribute the NSW Department of Primary Industries as the owner.

Disclaimer: The information contained in this publication is based on knowledge and understanding at the time of writing (June 2015). However, because of advances in knowledge, users are reminded of the need to ensure that information upon which they rely is up to date and to check currency of the information with the appropriate officer of the Department of Primary Industries or the user's independent adviser.

*Before July 2004, this report series was published by NSW Fisheries as the 'NSW Fisheries Final Report Series' with ISSN 1440-3544. Then, following the formation of the NSW Department of Primary Industries it was published as the 'NSW Department of Primary Industries – Fisheries Final Report Series' with ISSN 1449-9967. It was then published by Industry & Investment NSW as the 'Industry & Investment NSW – Fisheries Final Report Series' with ISSN 1837-2112. It is now published as the 'NSW Trade and Investment – Fisheries Final Report Series' with ISSN 1837-2112. It is now published as the 'NSW Trade and Investment – Fisheries Final Report Series' with ISSN 1837-2112.

Contents

Contents	. i
List of tables	ii
List of figures	ii
Acknowledgments	iv
Non-technical summary	v
Key words	.v
Introduction	1
Flood mitigation and degradation of Hexham Swamp	1
The Hexham Swamp Rehabilitation Project	1
Scope of this report	2
Methods	3
Experimental design	3
Fish and crustacean sampling	8
Water quality measurements	8
Hypothesis testing and statistical analyses 1	0
Reporting against OEMP detrimental impact1	1
Results1	2
Catch summary1	2
Assemblage changes associated with floodgate opening1	2
Water quality changes associated with floodgate opening1	7
Assessment of OEMP detrimental thresholds 1	7
Discussion	21
Impact of Ironbark Creek floodgate on Hexham Swamp2	21
Assemblage responses to the opening of Ironbark Creek floodgates	22
School and Eastern king prawn abundance2	23
Monitoring against OEMP trigger values2	25
Conclusions and recommendations2	26
References	27
Appendices	61
Appendix 1 – Catch summary	31
Appendix 2 – Prawn CPUE plots	32

List of tables

Table 1	Spatial experimental design applied on all sampling occasions	3
Table 2	Results of PERMANOVA analysis showing the effect of site and floodgate opening stage on assemblage composition	. 13
Table 3	Results of SIMPER analysis (Bray-Curtis dissimilarity, 4th-root transformed) showing the contribution that each of the 41 species made to the change in assemblage composition found at the upper Ironbark Creek site after Stage 3 opening of the floodgates. Species are ranked from biggest to smallest % contribution to dissimilarity. The change in mean catch per unit effort (CPUE: per four seines at a location) (untransformed) is shown to give an indication of the 'real' % change in abundance. Those species where the % change could not be defined (nd) were not present prior to Stage 3	. 14

List of figures

Figure 1	Spatial experimental design showing two replicate locations (each comprised of four collated seines) per site within each creek
Figure 2	Map of study area showing a) the location of Hexham Swamp relative to the lower Hunter River, and b) the location of study tidal creeks
Figure 3	Sampling locations within a) Ironbark Creek, b) Purgatory Creek, c) Cobbans Creek and d) Mosquito Creek
Figure 4	Sampling times throughout the 11-year study in relation to the staged opening of Ironbark Creek floodgates
Figure 5	Seine net being used to collect fish and crustacean samples
Figure 6	Preserved fish and crustacean samples after sorting in the laboratory
Figure 7	nMDS ordination showing a) assemblage differences between all sites at each time of sampling: plots b–d are the same ordination, but only show the trajectory of assemblage change for each site in Ironbark Creek; b) below the floodgate; c) at the lower site above the floodgate; and d) at the upper site above the floodgate. For ease of interpretation, ordination points have been removed for control and reference creeks in plots b-d and replaced with shading to represent the spread of points. The ordination was performed using Bray-Curtis similarity on 4th-root transformed data
Figure 8	Mean (\pm SE) species richness plots for each site at each phase of floodgate opening. From left to right are Ironbark, Purgatory and the combined reference creeks. From top to bottom are all, estuarine–marine (E–M), freshwater–estuarine (F–E) and commercially important (comm) species. Lines represent upper (dashed), lower (grey) and below-floodgate (black) sites. B = before; S1–S3 = Stage 1–3
Figure 9	Mean (\pm SE) water quality plots at each site at each phase of floodgate opening. From left to right are Ironbark, Purgatory and the combined reference creeks. From top to bottom is pH, salinity, dissolved oxygen and temperature. Lines represent upper (dashed), lower (grey) and below-floodgate (black) sites. B = before; S1–S3 = Stage 1–3
Figure 10	Change in mean species richness (freshwater–estuarine species excluded) in a) upper Ironbark Creek and b) lower Ironbark Creek sites at each sampling date throughout the study. The solid black lines shows the mean of the pre-opening

Acknowledgments

Bruce Pease was the original principal investigator of this study and devised the experimental design and coordinated sampling during the pre-opening phase. Many staff of the Department of Primary Industries (Fisheries NSW) assisted with field and laboratory work throughout this study, including Brooke McCartin, Tony Fowler, Gary Reilly, Ben Rampano, Ben Kearney, Brett Louden, Stephen Chilcott, Emma Mitchell, Brett Pflugrath and Emily Messer. Isabelle Thiebaud expertly prepared the maps. I am especially appreciative to the leaseholders of various sites used in the study for allowing staff access to sampling sites. Peggy Svoboda and other staff of the Kooragang Wetland Rehabilitation Project provided substantial assistance in accessing the sites on Ash Island.

This work was funded by the Hunter Local Land Services (formerly the Hunter-Central Rivers Catchment Management Authority). A number of Local Land Services staff provided substantial assistance, particularly Callaghan Cotter and Skye Moore in recent years. Members of the various community stakeholder groups are also acknowledged for their time and continued dedication to the health and recovery of Hexham Swamp and the lower Hunter River.

The collection of all animals in this project was done in accordance with the appropriate animal care and ethics research authority (98/11) and a Section 37 research permit in accordance with the NSW Fisheries Management Act 1994.

Non-technical summary

Changes in fish and crustacean assemblages in tidal creeks of Hexham Swamp following the staged opening of Ironbark Creek floodgates

Principal investigator:	Dr Craig A Boys
Address:	NSW Department of Primary Industries
	Port Stephens Fisheries Institute
	Locked Bag 1
	Nelson Bay NSW 2230
	Tel: 02 4982 1232. Fax: 02 4982 1107

Key words

Floodgates, Hunter River, tidal creek, tidal restriction, wetland rehabilitation

The Hunter River has experienced intensive flood mitigation over the past century, with 175 floodgates and hundreds of kilometres of levee banks and drainage canals being constructed. These developments have reduced the extent and biodiversity of the adjacent coastal wetlands such as Hexham Swamp. This Ramsar-listed wetland, located in the Hunter River estuary, was degraded by the installation of floodgates on Ironbark Creek in the early 1970s.

The Hexham Swamp Rehabilitation Project was established to promote the long-term rehabilitation of this important estuarine wetland ecosystem. The project has involved the staged opening of eight floodgates at the downstream end of Ironbark Creek to increase tidal flushing of the swamp and inundate low-lying lands with saline water. This involved three stages: one gate opened (Stage 1), three gates opened (Stage 2), six to eight gates opened (Stage 3).

This report documents the changing assemblage of aquatic fauna (fish and crustaceans) in Hexham Swamp over an 11-year period, between April 2004 and April 2014. It also outlines which of those changes, if any, may be attributed to the staged opening of the Ironbark Creek floodgates. To investigate the changes, seine netting was used to sample fish and crustaceans in tidal creeks above and below the Ironbark Creek floodgates. The catches were compared to nearby natural, un-gated reference creeks and a control creek that remained closed by a floodgate throughout the entire study.

Before the Ironbark Creek floodgates were opened, the following measurable impacts upstream of the floodgates were noted:

- reduced extent of suitable habitat for fish and crustaceans, with the upper site being less suitable than the lower site just upstream of the floodgate. This was reflected by the lower water quality in the upper site, including reduced pH (slightly acidic), dissolved oxygen and salinity
- lower species richness and abundance in the upper site, making the assemblage distinctly different from sites below the floodgates or in the other un-gated reference creeks

- far fewer estuarine-marine dwelling species in the upper reaches of the swamp, including the absence or substantially lower abundance of several commercially important species
- increased abundance of invasive Mosquitofish (Gambusia holbrooki) compared with un gated reference creeks.

After the floodgates were opened, significant recovery was noted in the upper site, which was more degraded than the lower site. Recovery was observed in both water quality and assemblage composition; by the end of the study, the upper site resembled the un-gated reference creeks. Most importantly, this response was not seen in the control creek where the floodgates remained closed. This provides strong evidence for the recovery being related to the opening of the floodgates. The recovery did not occur until all eight floodgates were opened in Stage 3.

Some notable responses in the upper site included:

- increased salinity, dissolved oxygen and pH, which all reached levels seen in un-gated creeks
- doubled species richness, including eight new species in Stage 3 alone. Most of the increase was seen in estuarine–marine dwelling species, but the number of freshwater– estuarine species also rose
- significantly increased abundance of many species, including several commercially important species
 - School prawn (*Metapenaeus macleayi*: 15 times more)
 - Eastern king prawn (*Melicertus plebejus*: absent prior to Stage 3)
 - Yellowfin bream (*Acanthopagrus australis*: 62 times more)
 - o Flat-tail mullet (Liza argentea: 10 times more)
 - Silver biddy (*Gerres subfasciatus*: 19 times more)
 - School prawn and Eastern king prawn were never sampled in the upper site prior to Stage 1 and Stage 3 opening (respectively), but since then were detected on every sampling occasion
- significantly reduced abundance of Mosquitofish.

On one occasion since the floodgates were opened, fish and crustacean species richness and abundance declined beyond the 75% trigger value that would denote a detrimental impact under the Hexham Swamp Rehabilitation Project's Operational and Environmental Monitoring Plan. This occurred in the upper Ironbark Creek site in July 2009, immediately following Stage 1 floodgate opening. It was a one-off observation, from which species richness and abundance recovered quickly.

In conclusion, the construction and closure of the Ironbark Creek floodgates significantly decreased the amount of suitable habitat for aquatic life within Hexham Swamp. As a result, the diversity and abundance of fish and crustaceans declined in tidal creeks. Opening the floodgates has reversed much of this impact, and restored connectivity between wetland and estuarine habitats for species across multiple levels in the food chain. It is therefore likely that ecosystem functioning and nutrient supply to the estuary has improved.

Significant improvements in species abundance and diversity were only seen once all eight gates were opened. This suggests that a threshold of increased tidal flushing, habitat improvement or water quality improvement may need to be exceeded before rehabilitation goals relating to fish and prawns can be realised.

The current study provides evidence to support the views of many commercial fishers – that installing the Ironbark Creek floodgates degraded important nursery habitat for species such as the school prawn and Eastern king prawn. The positive response of these and several other commercially important species to floodgate opening suggests that at least some of this impact

has been reversed. The degree to which this translates into improved productivity of commercial and recreational fisheries will depend on the contribution of those juveniles occupying the rehabilitated nursery habitats to the adult exploited population. This is the subject of ongoing research by NSW Department of Primary Industries.

Introduction

Flood mitigation and degradation of Hexham Swamp

Hundreds of floodgates, levee banks and drainage canals have been constructed along the Hunter River to mitigate occasional flooding (Williams and Watford 1997). Flood mitigation works of this degree have significantly altered the hydrology of the surrounding floodplain, degrading large areas of estuarine wetlands (Boys and Williams 2012a, b). Most notably, tidal flow into floodplain wetlands has been restricted, which typically changes the species assemblage and reduces biodiversity (Pressey and Middleton 1982, Herke et al. 1992, Pollard and Hannan 1994, Chambers et al. 1999, Kroon and Ansell 2006, Valentine-Rose et al. 2007, Eberhardt et al. 2011, Boys and Williams 2012b).

Hexham Swamp is the largest wetland in the Hunter River and one of the largest in New South Wales. It covers approximately 2000 ha around the Newcastle area. The ecosystem is internationally recognised under the Ramsar Convention due to its biological diversity and significance for migratory waterbirds.

In the early 1970s, as part of the Hunter Valley Flood Mitigation Scheme, floodgates were installed on Ironbark Creek – the main tributary of Hexham Swamp. This significantly reduced tidal inundation within Ironbark Creek and the creeks and marshes throughout Hexham Swamp (Haines 2011). Consequently, freshwater vegetation (mostly *Phragmites australis*) became dominant and saltmarsh and mangrove areas were replaced by freshwater reeds and pastures (Winning and Saintilan 2009).

The loss of estuarine habitat in Hexham Swamp also significantly reduced the diversity and abundance of estuarine fish (Genders 2001). Estuarine wetlands are important nurseries during early life stages of fish and invertebrate species, and contribute to the productivity of many estuarine and offshore fisheries (Morton 1990, Barbier and Strand 1998, Manson et al. 2005). Reports from commercial fishers indicate that Hexham Swamp once provided a major nursery for commercial species of fish and prawns (Simon Walsh, NSW DPI, unpublished data).

The Hexham Swamp Rehabilitation Project

The Hexham Swamp Rehabilitation Project was initiated for the long-term rehabilitation of the estuarine wetland ecosystem. As part of the project, floodgates located at the downstream end of Ironbark Creek were opened to increase tidal flushing of the swamp and periodically inundate low-lying lands with saline water. An important objective of the project was to re-establish estuarine wetland habitat and encourage recovery of estuarine wetland fauna.

The eight Ironbark Creek floodgates were opened in a staged manner, under the direction of a strict Operations and Environmental Management Plan (OEMP; Haines 2008). The primary reason for this staged approach was to adaptively manage any changes to the wetland, while maximising the opportunity for habitat and fauna recovery and minimising any negative environmental impacts (e.g. unexpected flooding, drops in water quality and associated fish kills, excessive erosion, mosquito outbreaks). The adaptive management framework involved regular monitoring of hydrology, water quality, vegetation, macroinvertebrates, birds, amphibians, fish and crustaceans.

Stage 1 of the Hexham Swamp Rehabilitation Project, which commenced on 19 December 2008, involved the partial opening of one floodgate. The resultant increase in tidal flows into and out of Hexham Swamp was modest, with only a small amount of overbank inundation (Haines 2009). The overall environmental condition within Ironbark Creek and Hexham Swamp appeared largely unaffected (Haines 2009).

Stage 2 of the project, which commenced one year later on 18 December 2009, involved the partial opening of three floodgates. This was equivalent to having two floodgates open completely, and resulted in the overbank inundation of 320 ha of land (BMT WBM 2011). Despite this increased inundation, no evidence was found that Stage 2 sufficiently increased tidal flushing in upstream sites within the swamp to improve fish and crustacean assemblages (Boys et al. 2012b).

Stage 3 commenced on 29 September 2011, when six floodgates were partially opened. The remaining floodgates were opened on 25 July 2013. All eight of the Ironbark Creek gates were then open to tidal flushing.

Scope of this report

This report documents the changing assemblage of fish and crustaceans in Hexham Swamp over an 11-year period (April 2004 to April 2014). It outlines which of those changes, if any, may be attributed to the staged opening of the Ironbark Creek floodgates. Of specific interest is whether estuarine–marine species (in this case, juvenile fish and crustaceans) have increased. The report also examines whether a detrimental threshold has been exceeded, as per the OEMP. Specifically, we sought to determine whether floodgate opening significantly reduced (by more than 75%) the abundance or diversity of aquatic species apart from those species expected to respond negatively, such as freshwater species (Haines 2008, 2009).

Methods

Experimental design

To evaluate the response to rehabilitation at Hexham Swamp, samples from Ironbark Creek (where floodgates were being opened or manipulated) were compared to reference creeks (not under the influence of floodgates) and an appropriate control creek (influenced by a floodgate that remained closed during the study) (Table 1, Figure 1 and Figure 2). This resulted in an unbalanced design consisting of one treatment or manipulated creek (Ironbark Creek), one control creek (Purgatory Creek) and two reference creeks (Cobbans and Mosquito creeks).

In each of the floodgated creeks, sampling was undertaken at a site immediately below the floodgate (referred to as 'below' sites) and at a site immediately above the floodgate ('lower' sites) (Figure 3). At Ironbark Creek, a site was also sampled in the upper reaches, known as Fisheries Creek ('upper' sites). Purgatory Creek, which is situated off the Hunter River upstream of Hexham Bridge, is too short for an upper site to be sampled, so only 'below' and 'lower' sites were sampled. The reference creeks are within the Kooragang Wetlands on Ash Island, and in each creek both an upper and lower site were sampled. Cobbans Creek is located off the south arm of the Hunter River, with its confluence with the Hunter River directly opposite that of Ironbark Creek (Figure 2). Mosquito Creek is located off the north arm of the Hunter River.

Fish and crustaceans were sampled at two replicate locations within each site (Table 1 and Figure 1) at various times before and after floodgate opening, during Stage 1, 2 and 3 (Figure 4). All 18 sites were sampled within a 1 to 2-week period during April, July, October and December.

Treatment	Creek	Site ^a	Replicate no. (location) ^b	Collated samples per location
Manipulated (floodgates opened during study)	Ironbark	Below floodgate Lower Ck above floodgate Upper Ck above floodgate	x 2 (IBB1, IBB2) x 2 (IBL1, IBL2) x 2 (IBU1, IBU2)	4 seines 4 seines 4 seines
Control (floodgate remained closed)	Purgatory	Below floodgate Lower Ck above floodgate	x 2 (PCB1, PCB2) x 2 (PCL1, PCL2)	4 seines 4 seines
Reference (no floodgates, remained tidally unrestricted during study)	Cobbans Mosquito	Lower Ck Upper Ck Lower Ck Upper Ck	x 2 (C2L1, C2L2) x 2 (C2U1, C2U2) x 2 (C6L1, C6L2) x 2 (C6U1, C6U2)	4 seines 4 seines 4 seines 4 seines

Table 1 Spatial experimental design applied on all sampling occasions

^a Refer to Figure 1 for schematic of spatial experimental design.

^b Location codes in parentheses relate to Figure 1 and Figure 3.

Figure 1 Spatial experimental design showing two replicate locations (each comprised of four collated seines) per site within each creek



Figure 2 Map of study area showing a) the location of Hexham Swamp relative to the lower Hunter River, and b) the location of study tidal creeks





Figure 3 Sampling locations within a) Ironbark Creek, b) Purgatory Creek, c) Cobbans Creek and d) Mosquito Creek

Refer to Table 1 for explanation of location codes.



Figure 4 Sampling times throughout the 11-year study in relation to the staged opening of Ironbark Creek floodgates

Fish and crustacean sampling

Fish and decapod crustaceans were collected during daylight hours. Each of the two locations within a site was sampled using four seine net hauls (10 m headline x 1.5 m drop x 3 mm stretch mesh), performed in a 'U'-shape and pursed onto the shore (Figure 5). Hauls were spaced no closer than ~10 m apart. A separate pilot study using up to eight seine hauls was performed on various occasions. This verified that four hauls were adequate in capturing the vast majority of species present, with species seldom added with subsequent hauling (Bruce Pease, NSW DPI, unpublished data). Other studies employing this sampling method in other estuaries of NSW have also found that three seine hauls captured 86% of species present at one location (Kroon and Ansell 2006). Seine netting began shortly before high tide and was completed shortly after to coincide with maximum depth and minimum velocity. The original design included fyke netting in upper sites. However, this was subsequently removed from the design, because it was very time consuming and the data obtained did not affect the interpretation of assemblage composition (Craig Boys, NSW DPI, unpublished data).

Fish and decapods were placed in buckets of estuarine water following capture. Fork length was recorded for fish and carapace length for decapod crustaceans. Length data was not analysed in this report, but is archived in a Fisheries NSW database. Larger individuals that could easily be identified to species level were processed on site and released alive. All other individuals were euthanased by lethal dose of Ethyl-p-amino benzoate (Benzocaine) (100 mg L-1) and each seine net catch was preserved in a separate labelled bag of formalin solution for later processing in the laboratory (Figure 6).

Water quality measurements

At the time of sampling, pH, dissolved oxygen, temperature and salinity were measured at the surface and bottom of the water column at each sampling site using a handheld Horiba U10 water quality meter. Samples were averaged across depths.



Figure 5 Seine net being used to collect fish and crustacean samples



Figure 6 Preserved fish and crustacean samples after sorting in the laboratory

Hypothesis testing and statistical analyses

The catches from four seine hauls at each site were pooled to give a single sample per location, per site within a creek. In total, 18 location samples (two locations per nine sites) were analysed per sampling occasion, totalling 360 samples from the 20 sampling occasions spread across the 11-year study.

A suite of exploratory multivariate techniques (within the PRIMER v6 statistical package) were used to determine whether the fish and crustacean assemblages within Ironbark Creek were affected by the presence of floodgates, and subsequently, whether any noticeable change occurred that could be attributed to the staged opening of the gates. Prior to multivariate analysis, the data were fourth-root transformed to ensure that the rarer, less abundant species also contributed to assemblage dissimilarity, rather than differences being dominated purely by the most abundant species (e.g. ambassids, eleotrids, shrimps) (Clarke and Green 1988).

Patterns in the fish and crustacean assemblage data were interpreted visually using non-metric multidimensional scaling ordination (nMDS; Kruskal and Wish 1978), based on Bray-Curtis dissimilarities between samples (Bray and Curtis 1957). Before dissimilarities were calculated, the average abundance of each species across the two locations at each site for each sampling occasion was calculated, resulting in one data point per site, per time of sampling. From the ordination, it was possible to track the change in assemblage composition at each site within Ironbark Creek over time, relative to the assemblages in the control and reference sites.

Nonparametric permutational analyses of variance (PERMANOVA; Type III sum-of squares) (Anderson 2001, McArdle and Anderson 2001) was used to determine if assemblage composition within Ironbark Creek sites changed significantly among the different stages of floodgate opening. If the test of main effects showed that a change had occurred, pairwise comparisons among each stage were subsequently performed to determine when these changes happened within each site. For upper Ironbark Creek sites (which showed the biggest assemblage response), the Similarity Percentages (SIMPER) procedure (Clarke 1993) was used to determine the relative contribution of the suite of species to the significant assemblage change identified by the PERMANOVA.

To examine changes in guilds of species based on their requirement to move between different parts of the estuary, the following functional groups were used (after Boys et al. 2012a).

- Estuarine-marine (E–M): saltwater species that require access to either estuarine or oceanic waters.
- Freshwater–estuarine (F–E): euryhaline species that can occupy both freshwater and saltwater.
- Freshwater (F) species: those typically confined to freshwater tributaries of estuaries.

Mean species richness (number) was calculated for these functional groups, as well as for total species and commercially important species. The results were presented as interaction plots to allow differences among sites and stages of floodgate opening to be compared. Mean interaction plots were presented in a similar way for the water quality parameters, to determine whether any changes may have been associated with floodgate opening.

Under a scenario where the Ironbark Creek floodgates had degraded the aquatic assemblage and water quality within Hexham Swamp, and gate opening rehabilitated the swamp, the following response was hypothesised. Before gate opening, the Ironbark Creek sites above the floodgate would differ from the sites below the floodgate and the two un-gated reference creeks. Once the floodgates were opened, the assemblage and/or water quality would be expected to become more similar to the sites below floodgates and in the un-gated reference creeks. A positive response to floodgate opening would be further supported if no such change was observed above the Purgatory Creek floodgate, which remained closed as a control.

Reporting against OEMP detrimental impact

As part of the OEMP, it was necessary to determine whether a detrimental threshold of a 75% reduction in species diversity or abundance was exceeded at any time following the opening of the floodgates. In this report, species richness (i.e., the total number of species) was reported rather than diversity. This was because preliminary investigation revealed that both richness and diversity (Shannon H') showed the same pattern of response, and richness is an easier concept to interpret than a relativised diversity measure.

Using a statistical control chart approach (Anderson and Thompson 2004), the mean change in species richness and abundance (Ln(N+1) transformed) was plotted for the upper and lower sites of Ironbark Creek. A significant change was concluded to have occurred if at any time the mean values fell outside of the 95% confidence interval for the pre-opening mean value. Changes were then judged against a 75% value of this mean pre-opening condition. Since the focus was on those species expected to benefit from floodgate opening, only estuarine–marine species were included in the analysis.

Results

Catch summary

In total, 54 fish species (136,433 individuals) and 14 decapod species (235,081 individuals) were sampled from tidal creeks in the lower Hunter River (Appendix 1). Ambassid, goby, shrimp and prawn were the most abundant types of taxa caught, with goby being the most diverse family (18 different species). Two-thirds (66%) of the species caught were primarily estuarine–marine dwelling, with the remaining third (34%) being species capable of inhabiting both freshwater and estuarine habitats. Close to a third (31%) of the species caught were juveniles of commercially important species, which accounted for 13% of the total abundance. The most abundant commercial species sampled were Flat-tail and Sea mullet (*Liza argentea* and *Mugil cephalus*); School, Greasyback and Eastern king prawn (*Metapenaeus macleayi, M. bennettae* and *Melicertus plebejus*) and Yellowfin bream (*Acanthopagrus australis*).

Assemblage changes associated with floodgate opening

In Ironbark Creek, the composition of the assemblage differed significantly among sites (upper, lower, below) and among the different stages of floodgate opening (PERMANOVA, Table 2). The assemblage at sites below and just above (lower sites) the floodgate varied through time, although by Stage 3 (with eight gates open) the composition had returned to the pre-opening stage (Table 2). Throughout the entire study, these two Ironbark Creek sites remained similar to control sites below Purgatory Creek floodgate and the un-gated reference sites (Figure 7b & c).

The major change observed in assemblage composition throughout the study was at the upper Ironbark Creek site. At this site the change did not occur until Stage 3, but it persisted, and by the end of the study the assemblage was significantly different from before floodgate opening (Table 2). The shift in assemblage composition was from one similar to that above the control floodgate at the start of the study, to one that had become similar to the un-gated reference creeks by the end (Figure 7d). Notably, this change in assemblage composition was not seen at the lower control site above Purgatory Creek floodgate, where the gates remained closed. At this site, the assemblage remained distinctly dissimilar in composition to that below the floodgate and that of the un-gated reference creeks throughout the entire study (Figure 7a).

The permanent shift in assemblage composition that occurred in the upper Ironbark Creek site once all eight floodgates were opened was partly driven by a significant increase in the number of species present (a doubling in richness, or eight new species in Stage 3 alone) (Figure 8). Most of the increase was in estuarine-marine dwelling species, but freshwater-estuarine species also increased. More commercially important species were also found. The additional species that were not present prior to Stage 3 are shown in Table 3, and notably included the commercially important Eastern king prawn. As a result of species addition, by the end of the study upper Ironbark sites had similar species richness to below-floodgate and reference sites. The change did not occur in these upper sites until all eight gates were opened, and a similar response was not observed in the control sites above the floodgate that remained closed (Purgatory Creek).

As well as more species being present in the upper Ironbark Creek site following the Stage 3 gate opening, many species changed significantly in abundance. SIMPER revealed that 25 of the 41 species found at this site explained 92.13% of the dissimilarity in assemblage composition between Stage 3 and the previous stages (Table 3). The catch per unit effort (CPUE) of each of these 25 species at each location and time before and after Stage 3 opening was consistent enough to suggest that they explained the assemblage change. Fifteen of these 25 species were estuarine–marine dwelling, and all of these increased in abundance. Ten species were freshwater–estuarine dwelling, and 50% of these declined in abundance. Included in the species that increased in abundance in the upper site were commercially important species, such as School prawn (15 times more), Eastern king prawn, Yellowfin bream (62 times

more), Flat-tail mullet (10 times more) and Silver biddy (*Gerres subfasciatus*) (19 times more). The five freshwater–estuarine species that decreased in abundance were Empire gudgeon (*Hypseleotris compressa*), Flathead gudgeon (*Philypnodon grandiceps*), Sea mullet, Pink shrimp (*Acetes sibogae australis*) and the invasive Mosquitofish (*Gambusia holbrooki*).

 Table 2 Results of PERMANOVA analysis showing the effect of site and floodgate opening stage on assemblage composition

Source	df	SS	MS	Pseudo-F	р			
Site Stage Site x stage Residual	2 3 6 47	24,859 13,602 7,649.6 41,489	12,429 4,534.2 1,274.9 882.8	14.08 5.1364 1.4443	0.001 0.001 0.015			
Summary of pairwise comparisons for factor Stage within site (p<0.05)								
Ironbark BelowBefore = Stage $3 \neq$ Stage $1 =$ Stage 2Ironbark LowerBefore = Stage $3 \neq$ Stage $1 \neq$ Stage 2Ironbark UpperBefore = Stage $1 =$ Stage $2 \neq$ Stage 3								

df = degrees of freedom; SS = sum of square; MS = mean squares; p = probability

Table 3 Results of SIMPER analysis (Bray-Curtis dissimilarity, 4th-root transformed) showing the contribution that each of the 41 species made to the change in assemblage composition found at the upper Ironbark Creek site after Stage 3 opening of the floodgates. Species are ranked from biggest to smallest % contribution to dissimilarity. The change in mean catch per unit effort (CPUE: per four seines at a location) (untransformed) is shown to give an indication of the 'real' % change in abundance. Those species where the % change could not be defined (nd) were not present prior to Stage 3

			SIMPER ^b		Mean CPUE [°]					
Common name	Scientific name ^a	Av. diss	Diss/SD	% cont.	B, S1 & S2	S 3	% change			
Grass shrimp	Palaemon debilis ^{E-M}	6.22	5.28	10	0.7	289.3	41.229			
School prawn	Metapenaeus macleavi E-M d	5.19	2.23	8.36	18.2	292.6	1,508			
Swan River goby	Pseudogobius olorum	4.44	3.12	7.14	1.9	122.9	6,368			
Largemouth goby	Redigobius macrostoma ^{E–M}	4.14	3.2	6.66	1.3	94.6	7,177			
Glass goby	Gobiopterus semivestitus ^{E–M}	3.55	1.71	5.71	46.5	185.6	299			
Pink shrimp	Acetes sibogae australis ^{F–E}	3.47	1.5	5.58	237.2	147.8	-38			
Port Jackson glassfish	Ambassis jacksoniensis ^{E–M}	2.82	1.4	4.54	29.4	48.9	66			
Yellowfin bream	Acanthopagrus australis ^{E-M d}	2.58	2.58	4.15	0.2	12.6	6,200			
Glassfish	Ambassis spp $E-M$	2.58	1.44	4.15	1.5	32.8	2,087			
Striped gudgeon	Gobiomorphus australis ^{F–E}	2.35	1.53	3.77	12.0	98.4	720			
Mangrove goby	Mugilogobius paludis ^{E–M}	2.03	2.04	3.26	0.3	7.3	2,333			
Sea mullet	Mugil cephalus ^{F-E d}	1.66	1.22	2.66	33.8	1.8	-95			
Southern blue-eye	Pseudomugil signifer ^{F-E}	1.64	1.29	2.64	4.1	12.8	212			
Mosquitofish	Gambusia holbrooki ^{F–E}	1.62	1.3	2.61	39.9	14.9	-63			
Flat-tail mullet	Liza argentea ^{E-M d}	1.59	1.08	2.56	0.8	8.8	1,000			
Empire gudgeon	Hypseleotris compressa ^{F–E}	1.54	1.36	2.47	30.7	12.4	-60			
Silver biddy	Gerres subfasciatus E-M d	1.51	1.43	2.43	0.1	2.0	1,900			
Eastern king prawn	Melicertus plebejus ^{E-M d}	1.38	0.98	2.22	0.0	4.6	nd			
Tamar River goby	Afurcagobius tamarensis F-E	1.31	1.11	2.11	0.9	3.4	278			
Flathead gudgeon	Philypnodon grandiceps ^{F–E}	1.24	1.56	1.99	17.9	2.3	-87			
Pistol shrimp	Alpheus spp ^{Ē_M}	1.04	0.98	1.67	0.0	1.0	nd			
Grapsid crab	Parasesarma erythrodactyia ^{E–M}	0.88	0.97	1.42	0.0	0.5	nd			
Half-bridled goby	Arenigobius frenatus ^{E–M}	0.87	0.94	1.39	0.0	0.5	nd			
Striped shrimp	Macrobrachium intermedium F-E	0.83	0.98	1.34	0.0	0.8	nd			
Eastern fortescue	Centropogon australis ^{E-M}	0.81	0.96	1.31	0.0	0.9	nd			
Estuary perchlet	Ambassis marianus ^{E–M}	0.64	0.62	1.03	0.3	1.0	233			
Exquisite sand goby	Favonigobius exquisitus ^{E–M}	0.48	0.63	0.78	0.2	0.1	-50			
Greasyback prawn	Metapenaeus bennettae	0.47	0.68	0.76	0.3	0.1	-67			
Sand whiting	Sillago ciliata E-M d	0.45	0.63	0.73	0.1	0.1	0			
Australian bass	Percalates novemaculeata F-Ed	0.45	0.57	0.72	0.0	0.6	nd			
Tarwhine	Rhabdosargus sarba ^{E–M d}	0.44	0.63	0.71	0.03	0.4	1,233			
Dwarf flathead gudgeon	Philypnodon macrostomus F-E	0.44	0.65	0.71	0.3	0.0	-100			
Shrimp	Palaemonidae spp ^{E⊣M}	0.38	0.56	0.62	0.3	0.0	-100			
Bridled goby	Arenigobius bifrenatus ^{E-M}	0.34	0.57	0.55	0.0	0.13	nd			
Goldfish	Carassius auratus ^{F–E}	0.2	0.25	0.32	0.7	0.0	-100			
Short-finned eel	Anguilla australis ^{F–E d}	0.16	0.37	0.26	0.1	0.0	-100			
Fire Tailed gudgeon	Hypseleotris galii ^{F–E}	0.1	0.26	0.15	0.1	0.0	-100			
Long-finned eel	Anguilla reinhardtii ^{F–E d}	0.09	0.25	0.14	0.03	0.0	-100			
Golden goby	Glossogobius biocellatus ^{E–M}	0.08	0.25	0.13	0.03	0.0	-100			
Tailor	Pomatomus saltatrix ^{E-M d}	0.08	0.25	0.13	0.03	0.0	-100			
False spider crab	Amarinus lacustris ^{F-E}	0.08	0.26	0.12	0.03	0.0	-100			

a E-M = estuarine–marine, F-E = freshwater–estuarine; **b** 4th-root transformed; **c** untransformed; **d** = commercially important. N.B. Only those species appearing above the dashed line have sufficiently high consistency ratios (Diss/SD) to suggest that they consistently contribute to dissimilarity, i.e. those species below the dashed line are not good discriminators.

Figure 7 nMDS ordination showing a) assemblage differences between all sites at each time of sampling: plots b-d are the same ordination, but only show the trajectory of assemblage change for each site in Ironbark Creek; b) below the floodgate; c) at the lower site above the floodgate; and d) at the upper site above the floodgate. For ease of interpretation, ordination points have been removed for control and reference creeks in plots b-d and replaced with shading to represent the spread of points. The ordination was performed using Bray-Curtis similarity on 4th-root transformed data





20

16

Stage 3

Figure 8 Mean (±SE) species richness plots for each site at each phase of floodgate opening. From left to right are Ironbark, Purgatory and the combined reference creeks. From top to bottom are all, estuarine-marine (E-M), freshwater-estuarine (F-E) and commercially important (comm) species. Lines represent upper (dashed), lower (grey) and below-floodgate (black) sites. B = before; S1-S3 = Stage 1-3



Water quality changes associated with floodgate opening

Water quality improved in Ironbark Creek as a result of floodgate opening. The biggest response was seen at the upper Ironbark Creek site, where pH, salinity and dissolved oxygen all increased (Figure 9). While pH and dissolved oxygen improved with every subsequent stage of floodgate opening, salinity did not improve in upper sites until Stage 3. The increase in salinity at this site during Stage 3 was large, changing from between 10 and 15 ppt to ~25 ppt following opening: comparable to un-gated reference creeks. The upper Ironbark Creek site was also the only site found to be acidic, although this was only before floodgate opening.

In contrast to the upper Ironbark Creek site, the water quality at the lower Ironbark Creek site appeared less affected by the floodgate, and therefore remained similar to comparably located sites in the reference creeks throughout all stages. An upstream pH and salinity gradient before floodgate opening was below > lower > upper, but this gradient was significantly reduced by Stage 3. The main impact observed at Purgatory Creek was a slight lowering of salinity upstream of the floodgate during most stages of the study.

In comparison with the other water quality parameters, water temperature appeared minimally affected by floodgate presence or opening. Changes in water temperature over time reflected estuary-scale changes that were consistent at all sites throughout the study.

Assessment of OEMP detrimental thresholds

Species richness fell below the 75% detrimental reduction level on only one occasion (Figure 10). This occurred at the upper Ironbark Creek site on the first sampling occasion after the initial Stage 1 opening. With the exception of one occasion in Stage 1, it was not until Stage 3 that species richness in the upper site was significantly greater than before floodgate opening. Consistent with previous reports, species richness did not significantly increase at the lower site.

The abundance of estuarine–marine species also fell below the 75% detrimental reduction trigger value on one occasion (Figure 11 and Figure 10). As with species richness, this occurred at the upper Ironbark Creek site on the first sampling occasion after the initial Stage 1 opening. Although abundance varied substantially over time, a significant increase in abundance was only sustained for consecutive sampling times at the upper site during Stage 3.

Figure 9 Mean (±SE) water quality plots at each site at each phase of floodgate opening. From left to right are Ironbark, Purgatory and the combined reference creeks. From top to bottom is pH, salinity, dissolved oxygen and temperature. Lines represent upper (dashed), lower (grey) and belowfloodgate (black) sites. B = before; S1–S3 = Stage 1–3



Figure 10 Change in mean species richness (freshwater–estuarine species excluded) in a) upper Ironbark Creek and b) lower Ironbark Creek sites at each sampling date throughout the study. The solid black lines shows the mean of the pre-opening values and the dashed lines the upper and lower 95% confidence intervals of that mean. The 75% reduction threshold as identified in the OEMP is shown



a) Iron bark Creek upper site

Survey date

Figure 11 Change in mean catch per unit effort (CPUE) [Ln(N+1) transformed] (freshwater–estuarine species excluded) in a) upper Ironbark Creek and b) lower Ironbark Creek sites at each sampling date throughout the study. The solid black lines shows the mean of the pre-opening values and the dashed lines the upper and lower 95% confidence intervals of that mean. The 75% reduction threshold as identified in the OEMP is shown



a) Iron bark Creek upper site

Survey date

Discussion

Impact of Ironbark Creek floodgate on Hexham Swamp

Prior to the Stage 1 opening of the Ironbark Creek floodgates, the assemblage at the lower site immediately above the floodgate was not significantly different from the assemblage below the floodgate or in lower sites of reference creeks. This suggests that the gates had either been opened at some stage to allow fish passage, or display a certain degree of 'leakiness', as observed at floodgates in other estuaries (Kroon and Ansell 2006).

Despite the evidence of passage in the immediate vicinity of the Ironbark Creek gates prior to the Stage 1 opening, the gates had clearly been affecting the assemblage over larger spatial scales in Hexham Swamp. The upper Ironbark Creek site was quite distinct in assemblage composition from the lower and below gate sites, primarily driven by a lower abundance of estuarine–marine dwelling species at the upper site. This study adds to the growing body of evidence showing that reduced tidal flushing of coastal wetlands can significantly reduce species diversity (e.g. Raposa and Roman 2003, Eberhardt et al. 2011, Boys et al. 2012a, Boys and Williams 2012a).

A reduction in estuarine–marine species was not associated with greater numbers of freshwater species. Entirely freshwater species were absent in tidal creeks within Hexham Swamp, as has been reported in floodgated creeks in the Macleay and Clarence Rivers (Boys et al. 2012a). Instead, floodgated creeks seem to be favoured by species that can tolerate the large variations in salinity levels that can occur in tidally restricted wetlands.

A reduction in species in aquatic assemblages can affect food webs and ecosystem functioning (Petchey et al. 2004). For example, species such as the Pistol shrimp (*Alpheus* spp.) and Striped shrimp (*Macrobrachium intermedium*) form important food sources for fish and wading birds (Miranda and Collazo 1997), but their distribution was restricted to the lower site prior to floodgate opening. Similar effects on shrimp species have been reported in floodgated creeks elsewhere (e.g. *Acetes sibogae australis*: Kroon and Ansell 2006, Boys et al. 2012a). Many other estuarine–marine species were absent or in much reduced abundance within Hexham Swamp before the floodgates were opened. In a general sense, restricting the movement of fish into wetlands can have ramifications for the larger coastal marine ecosystem, because fish migration can play an important role in exporting the productivity of wetlands (e.g. carbon and nitrogen) to estuaries and the marine environment (Deegan 1993).

Although floodgates pose a physical barrier to fish passage for at least a large proportion of time, many of the species impacts observed in Hexham Swamp may have been caused by reduced water quality and estuarine habitat values upstream of the floodgates. Numerous examples in the literature describe how floodgates can degrade estuarine habitats, particularly the physicochemical environment in which these species live (Roman et al. 1984, Pollard and Hannan 1994, Kroon and Ansell 2006, Boys et al. 2011). By restricting the tidal flushing of Ironbark Creek, the floodgates significantly decreased the area of occupiable habitat within Hexham Swamp, making the upper site less suitable and restricting species distribution to the lower site just above the floodgate.

On average, the pH was acidic in the upper site prior to floodgate opening. Acidification can result when coastal wetlands with acid sulphate soils are floodgated and artificially drained (Sammut et al. 1996): a scenario common throughout the lower Hunter River floodplain. Chronic acidification make habitats less suitable for species to occupy and affect fish growth and reproductive behaviour (Beamish et al. 1975, Sammut et al. 1995). Episodic acidification events with a rapid onset can also result in fish kills and predispose fish to fatal bacterial and fungal diseases, such as red spot disease (Sammut et al. 1995, Sammut et al. 1996, Sanaullah et al. 2001). These effects are amplified when lower pH is combined with other sub-optimal water

quality parameters, such as lower dissolved oxygen and salinity (Lilley et al. 1998), which also compromise fish immunity – as was observed in Ironbark Creek.

Assemblage responses to the opening of Ironbark Creek floodgates

The opening of Ironbark Creek floodgates coincided with a significant improvement in both species diversity and abundance. Positive responses mainly involved estuarine-marine dwelling species, which indicates that connectivity with the estuary is an important factor mediating the presence of this guild in tidal wetlands. Benefits were observed across multiple trophic levels, from detritivores such as shrimps and prawn (e.g. Grass shrimp, *Palaemon debilis*; School prawn and Eastern king prawn) to opportunistic generalists that feed on small invertebrates and fish (e.g. Yellowfin bream and a variety of gobiid species).

Increasing the diversity and abundance of species across multiple trophic levels will improve the complexity of food webs. In turn, this can increase ecosystem function and resilience to buffer ecosystems against disturbance (Hooper et al. 2005, Fischer et al. 2006). By maintaining connectivity between tidally active wetlands and the rest of the estuary, biota can move freely throughout the landscape and ensure that resource, genetic and process-based linkages are maintained (Lundberg and Moberg 2003).

Ironbark Creek improved as potential nursery habitat for commercially important species once the floodgates were opened, with more juvenile Yellowfin bream, Flat-tail mullet, Silver biddy, School prawn and Eastern king prawn being sampled. Connectivity between juvenile and adult habitats is critical to the value of estuarine and marine nurseries (Beck et al. 2001). Benefits gained from additional nursery habitat in tidal wetlands are expected to translate into greater productivity of commercial and recreational estuarine and offshore fisheries. Studies investigating long-term fisheries data from the subtropics demonstrate that catches tend to be highest in estuaries of greatest tidal wetland habitat availability and connectivity (Turner 1992, de Graaf and Xuan 1999, Manson et al. 2005a, Manson et al. 2005b, Meynecke et al. 2008, Meynecke 2009).

Floodgate opening was also associated with reduced abundance of the invasive Mosquitofish in Hexham swamp. This supports the growing body of knowledge suggesting that reduced hydrological connection between wetlands and marine waters favours populations of invasive species, particularly *Gambusia* (Kroon and Ansell 2006, Boys and Williams 2012b, MacKenzie and Bruland 2012). Importantly, this study demonstrates that reinstatement of tidal flushing can measurably reduce the number of Mosquitofish. Such improvements are beneficial for wetland ecosystems, because in large numbers, Mosquitofish can alter food webs and affect nutrient cycling through heavy feeding on rotifer, crustacean, and insect populations, which subsequently increases phytoplankton populations (Hurlbert et al. 1972, Hurlbert and Mulla 1981). Mosquitofish also prey heavily on frog eggs and tadpoles (Grubb 1972, Baber et al. 2004), including the eggs and tadpoles of the endangered green and golden bell frog (*Litoria aurea*) (Pyke and White 2000). This nationally protected frog species co-inhabits wetland habitats in the Hunter River estuary (Hamer et al. 2002).

The staged manner in which the eight floodgates were opened gave us a rare opportunity to investigate the extent of water quality and assemblage change in Ironbark Creek under differing degrees of tidal flushing. Even with the gates closed, water quality appeared to be minimally affected in the lower site compared with the upper site. Consequently, assemblage and water quality changes were not pronounced in the lower site following floodgate opening. In comparison, the upper site was substantially affected by sub-optimal water quality and reduced species diversity and abundance prior to opening. These metrics significantly improved once the floodgates were opened.

The improvement in dissolved oxygen and pH was incremental, coinciding with each stage of opening, whereas salinity did not improve until the final stage. It was not until Stage 3 that water quality was similar to un-gated reference creeks. Similarly, it was not until Stage 3 that assemblage composition was similar to un-gated reference creeks. These findings support those of others, who have shown that fish and crustacean assemblage responses to wetland restoration will be greatest in wetlands that are much more tidally restricted than other wetlands (Raposa and Roman 2003). It also illustrates that floodgate opening and improved passage on its own may not be sufficient to rehabilitate wetland assemblages, unless a threshold of tidal flushing is exceeded to allow sufficient improvement in water quality and habitat condition. In the case of Ironbark Creek, this meant opening all gates, rather than one or three. In a nearby creek on Ash Island in the lower Hunter (Fish Fry Creek), eight years passed after culvert removal before the creek's hydrogeology and geomorphology had changed sufficiently to support a fish and crustacean assemblage similar in composition to nearby, natural, unrestricted creeks (Boys and Williams 2012a).

School and Eastern king prawn abundance

This monitoring program never intended to target or describe the responses of specific species to floodgate opening. However, there has been particular community interest in the responses of two commercially important penaeid prawn species: Eastern king prawn and School prawn.

Hexham Swamp was historically believed to be a nursery for juvenile prawns. Oral accounts from local commercial prawn fishers indicate that the swamp used to supply the lower Hunter with significant numbers of both Eastern king and School prawn recruits (Simon Walsh 2014, NSW DPI, unpublished information). If this was the case, then Hexham Swamp may have been contributing significantly to the productivity of both the lower Hunter prawn fishery and the broader east coast of Australia. Tagging studies have already confirmed that Eastern king prawns from NSW estuaries (including the lower Hunter River) can migrate large distances north to spawning grounds off the Queensland coast, thus contributing to a stock exploited by estuarine and offshore fisheries (Ruello 1975, Montgomery 1990). It may therefore not be surprising that commercial fishers believe the productivity of the Hunter River prawn fishery reduced dramatically after the Ironbark Creek floodgates were installed.

Although the oral accounts of commercial fishers are compelling, limited data is available on the potential use of tidal creeks by penaeids in NSW estuaries. Eastern king prawn are known to associate with seagrass (Ochwada-Doyle et al. 2009), but this habitat is lacking in the Hunter River estuary. Large numbers of Eastern king prawn have been found at channel edges near mangrove and saltmarsh in several rivers, including the Hunter (Gibbs et al. 1999). This supports studies from subtropical Queensland, which highlight the potential importance of mangrove-lined wetlands as nurseries for this species (Young and Carpenter 1977, Young 1978 Halliday 1995, Skilleter et al. 2005). Unlike the Eastern king prawn, the use of tidal creeks by juvenile School prawn has been clearly demonstrated in recent studies (e.g. Boys et al. 2012a, Boys and Williams 2012a). School prawn is a euryhaline species that is distributed well into the upper Hunter River. They have a greater tolerance for lower salinities than that of the Eastern king prawn, which may make tidal creeks more suitable for the School prawn (Dall 1981).

The sampling techniques used in this study may not have been ideal for targeting penaeid prawns, since night sampling was not performed (Guest et al. 2003). Despite this, we obtained large numbers of prawns in the seine nets at times. Applying standardised sampling methods throughout the 11-year study has allowed us to compare the relative changes in prawn distribution and abundance following floodgate opening in the lower Hunter and Hexham Swamp.

Juvenile Eastern king prawn and School prawn occupy tidal creeks in the lower Hunter estuary in a seasonal manner (Figures A2.1 and A2.2 in Appendix 2). Juvenile Eastern king prawn abundances in tidal creeks peaked in spring (October), and tended to fall off rapidly, so that by summer (December) they were rarely sampled in most years. This supports the limited amount of research from temperate waters of east coast Australia, which suggest the main recruitment of Eastern king prawn post larvae to estuaries occurs between September and October (Racek 1959). After about three months in nursery habitats, they run out to sea to mature and mate (Racek 1959, Young 1975). A 'short' nursery estuarine phase is also supported by oral accounts of commercial prawn fishers, some of whom target schools on their out-migration (Simon Walsh 2014, NSW DPI, unpublished information). School prawn appeared to have slightly longer residency in tidal creeks of the Hunter: peaking in abundance in December, but being sampled in smaller numbers throughout the year.

As well as seasonal variation in prawn abundance, there were also clear differences between years (Figures A2.1 and A2.2 in Appendix 2). Of particular note was the strong recruitment of Eastern king prawns to the Hunter estuary in the spring of 2013. The supply of post-larvae to estuaries on the NSW coast is likely to be driven by extraneous factors, such as relative spawning success and the characteristics of the Eastern Australian Current (Montgomery 1990). Whether this or other oceanographic factors drove the 2013 abundance peak is outside the scope of this study. It does, however, highlight the potential benefit arising from the reinstatement of estuarine connectivity with Hexham Swamp.

It is clear that with the Ironbark Creek floodgates closed, less nursery habitat would have been available for both Eastern king prawn and School prawn. Before Ironbark Creek was opened, both species were rarely caught in Hexham Swamp, and their distribution was limited to a few individuals in lower reaches near the floodgate. Following Stage 1 floodgate opening, significantly more School prawns occupied the lower reaches of Ironbark Creek; after Stage 2 opening, School prawns were being sampled from upper Ironbark Creek for the first time since the study began. By Stage 3, 15 times more School prawns were being sampled from the upper site, on average, compared with previous stages (Table 3). A similar response was seen for Eastern king prawn, which were detected in significantly higher numbers in lower sites following Stage 2 opening, and penetrated into the upper site for the first time following opening of all floodgates in Stage 3. This response was undoubtedly mediated by improvements in both passage and water quality. Salinity in the upper site only increased to around 25 ppt after Stage 3, bringing it closer to the optimal range for juvenile Eastern king prawn (28–30 ppt; Dall 1981).

When peak abundances of School and Eastern king prawn were sampled in the Hunter estuary, prawns were prevented from accessing tidal creek habitats upstream of the Purgatory Creek and Ironbark Creek floodgates before Stage 1 opening. As more floodgates were opened at Ironbark Creek and tidal flushing extended further into Hexham Swamp, the extent of suitable prawn habitat also increased. Clearly, floodgates at Hexham Swamp degraded prawn habitat in the Hunter River; but, most importantly, proper floodgate management has significantly increased the extent of available habitat. This is a significant finding, because it is the first time an increase in prawn occupancy in tidal creeks has been detected following floodgate opening or culvert removal in NSW estuaries. The removal of tidal restrictions in several NSW estuaries (e.g. Hunter, Clarence, Macleay) has often led to a rapid and sustained improvements in fish and decapod diversity and abundance. However, an increase in School or Eastern king prawn abundance had not been noted until now (Boys et al. 2012a, Boys and Williams 2012a).

While the opening of Ironbark Creek floodgate has increased the availability of prawn habitat in the lower Hunter River, the degree to which such changes will contribute to improved commercial productivity is still uncertain. A study is currently underway by NSW Department of Primary Industries involving stable isotope analysis and quantitative sampling of tidal creeks for

prawns. For the first time, this study should provide an indication of the nursery contribution made by un gated tidal creeks and recently rehabilitated creeks such as Ironbark Creek.

Monitoring against OEMP trigger values

On only one occasion did fish and decapod species richness and abundance decline beyond the 75% trigger value, which would denote a detrimental impact to floodgate opening under the OEMP. This occurred in the upper Ironbark Creek site in July 2009, immediately following Stage 1 floodgate opening. It was a one-off observation, from which species richness and abundance recovered quickly.

The episodic nature of the decline, along with its occurrence close to the time of floodgate opening, suggests that it was unlikely to be caused by tidally induced effects on swamp habitat (e.g. freshwater vegetation die-off). Instead, it was likely to be caused by significant amounts of rain and higher-than-usual freshwater inputs to the swamp, which resulted in fewer estuarine-marine species being collected.

Conclusions and recommendations

The construction and closure of the Ironbark Creek floodgates significantly decreased the amount of suitable habitat for aquatic biota within Hexham Swamp, contributing to a decline in the number and abundance of estuarine-marine fish and crustacean species. The opening of the Ironbark Creek floodgates has altered the species assemblage within Hexham Swamp, although significant changes only occurred after all eight gates were opened. Such a threshold response is not surprising, given that having all gates opened most closely mimics the natural, un-gated condition. This suggests that management of floodgates or other tidal restrictions for wetland rehabilitation needs to recognise that ecological responses may not follow until a certain degree of wetland flushing and water quality or habitat availability is achieved.

The current study corroborates the views of many commercial fishers that the installation of the Ironbark Creek floodgates degraded important nursery habitat for species such as the Eastern king prawn. The positive response of this and several other commercially important species to floodgate opening suggests that improved recruitment of these species is a likely outcome of wetland rehabilitation. The degree to which these positive changes translate into improved productivity of commercial and recreational fisheries will depend on the contribution of those juveniles occupying rehabilitated nursery habitats to the adult exploited population. This is the subject of ongoing research by NSW Department of Primary Industries.

References

- Anderson, M. J. (2001). A new method for non-parametric multivariate analysis of variance. *Austral Ecology* **26**: 32-46.
- Anderson, M. J. and Thompson, A. A. (2004). Multivariate control charts for ecological and environmental monitoring. *Ecological Applications* **14**: 1921-1935.
- Baber, M. J., Babbitt, K. J. and Douglas, M. E. (2004). Influence of habitat complexity on predator–prey interactions between the fish (*Gambusia holbrooki*) and tadpoles of *Hyla* squirella and Gastrophryne carolinensis. Copeia **2004**: 173-177.
- Barbier, E. B. and Strand, I. (1998). Valuing mangrove-fishery linkages. A case study of Campeche, Mexico. *Environmental Resource Economics* **12**: 151-166.
- Beamish, R., Lockhart, W., Van Loon, J. and Harvey, H. (1975). Long-term acidification of a lake and resulting effects on fishes. *Ambio*: 98-102.
- Beck, M. W., Heck Jr, K. L., Able, K. W., Childers, D. L., Eggleston, D. B., Gillanders, B. M., Halpern, B., Hays, C. G., Hoshino, K. and Minello, T. J. (2001). The identification, conservation, and management of estuarine and marine nurseries for fish and invertebrates. *BioScience* **51**: 633-641.
- BMT WBM (2011). Hexham Swamp Rehabilitation Project: Stage 2 Environmental Performance Assessment Report., Broadmeadow.
- Boys, C. A., Glasby, T., Kroon, F. J., Baumgartner, L. J., Wilkinson, K., Reilly, G. and Fowler, T. (2011). Case studies in restoring connectivity of coastal aquatic habitats: floodgates, box culvert and rock-ramp fishway. Fisheries Final Report Series No. 130. NSW Department of Primary Industries, Cronulla, 70 pp.
- Boys, C. A., Kroon, F. J., Glasby, T. and Wilkinson, K. (2012a). Improved fish and crustacean passage in tidal creeks following floodgate remediation. *Journal of Applied Ecology* 49: 223-233.
- Boys, C. A., Pease, B. and Robinson, W. (2012b). Responses of fish and crustacean assemblages in Hexham Swamp following Stage 1 & 2 opening of Ironbark Creek floodgates. Report to the Hunter-Central Rivers CMA. NSW Department of Primary Industries, 38 pp.
- Boys, C. A. and Williams, R. J. (2012a). Succession of fish and crustacean assemblages following reinstatement of tidal flow in a temperate coastal wetland. *Ecological Engineering* **49**: 221-232.
- Boys, C. A. and Williams, R. J. (2012b). Fish and decapod assemblages in Kooragang Wetlands: the impact of tidal restriction and responses to culvert removal. Fisheries Final Report Series No. 133. NSW Department of Primary Industries, Cronulla, 80 pp.
- Bray, J. R. and Curtis, J. T. (1957). An ordination of the upland forest communities of Southern Wisconsin. *Ecological Monographs* **27**: 325-349.
- Chambers, R. M., Meyerson, L. A. and Saltonstall, K. (1999). Expansion of Phragmites australis into tidal wetlands of North America. *Aquatic Botany* **64**: 261-273.
- Clarke, K. R. and Green, R. H. (1988). Statistical design and analysis for a 'biological effects' study. *Marine Ecology Progress Series* **46**: 213-226.
- Clarke, K. R. (1993). Non-parametric multivariate analyses of changes in community structure. *Australian Journal of Ecology* **18**: 117-143.
- Dall, W. (1981). Osmoregulatory ability and juvenile habitat preference in some penaeid prawns. *Journal of Experimental Marine Biology and Ecology* **54**: 55-64.

- Deegan, L. A. (1993). Nutrient and Energy Transport between Estuaries and Coastal Marine Ecosystems by Fish Migration. *Canadian Journal of Fisheries and Aquatic Sciences* **50**: 74-79.
- Eberhardt, A. L., Burdick, D. M. and Dionne, M. (2011). The effects of road culverts on nekton in New England salt marshes: Implications for tidal restoration. *Restoration Ecology* **19**: 776-785.
- Fischer, J., Lindenmayer, D. B. and Manning, A. D. (2006). Biodiversity, ecosystem function, and resilience: ten guiding principles for commodity production landscapes. *Frontiers in Ecology and the Environment* **4**: 80-86.
- Genders, A. J. (2001). Distribution and abundance of larval fishes and invertebrates along the Hunter River estuary, New South Wales, Australia, with specific reference to the effects of floodgates. University of Newcastle, Newcastle. 217 pp.
- Grubb, J. C. (1972). Differential predation by *Gambusia affinis* on the eggs of seven species of anuran amphibians. *American Midland Naturalist* **88**: 102-108.
- Guest, M. A., Connolly, R. M. and Loneragan, N. R. (2003). Seine nets and beam trawls compared by day and night for sampling fish and crustaceans in shallow seagrass habitat. *Fisheries Research* **64**: 185-196.
- Haines, P. (2008). Hexham Swamp Rehabilitation Project Operations Environmental Management Plan. BMT WBM, Newcastle, 209 pp.
- Haines, P. (2009). Hexham Swamp Rehabilitation Project Stage 1: Environmental Performance Assessment Report. BMT WBM, Newcastle.
- Haines, P. (2011). Hexham Swamp Rehabilitation Project Stage 2: Environmental Performance Assessment Report. Report to the Hunter Local Land Services (formerly the Hunter-Central Rivers Catchment Management Authority). BMT WBM, Newcastle, 91 pp.
- Halliday, I. A. (1995). Influence of natural fluctuations in seagrass cover on commercial prawn nursery grounds in a subtropical estuary. *Marine and Freshwater Research* **46**: 1121-1126.
- Hamer, A. J., Lane, S. J. and Mahony, M. J. (2002). Management of freshwater wetlands for the endangered green and golden bell frog (*Litoria aurea*): roles of habitat determinants and space. *Biological Conservation* **106**: 413-424.
- Herke, W. H., Knudsen, E. E., Knudsen, P. A. and Rogers, B. D. (1992). Effects of semiimpoundment of Louisiana marsh on fish and crustacean nursery use and export. *North American Journal of Fisheries Management* **12**: 151–160.
- Hooper, D., Chapin Iii, F., Ewel, J., Hector, A., Inchausti, P., Lavorel, S., Lawton, J., Lodge, D., Loreau, M. and Naeem, S. (2005). Effects of biodiversity on ecosystem functioning: a consensus of current knowledge. *Ecological monographs* **75**: 3-35.
- Hurlbert, S. H., Zedler, J. and Fairbanks, D. (1972). Ecosystem alteration by mosquitofish (*Gambusia affinis*) predation. *Science* **175**: 639.
- Hurlbert, S. H. and Mulla, M. S. (1981). Impacts of mosquitofish (*Gambusia affinis*) predation on plankton communities. *Hydrobiologia* **83**: 125-151.
- Kroon, F. J. and Ansell, D. H. (2006). A comparison of species assemblages between drainage systems with and without floodgates: implications for coastal floodplain management. *Canadian Journal of Fisheries and Aquatic Sciences* 63: 2400-2417.
- Kruskal, J. B. and Wish, M. (1978) Multidimensional Scaling. Sage Publications, London. 93 pp.
- Lilley, J. H., Callinan, R. B., Chinabut, S., Kanchanakhan, S., MacRae, I. H. and Phillips, M. J. (1998). *Epizootic ulcerative syndrome (EUS) technical handbook*. Aquatic Animal Health Research Institute, Bangkok, 88 pp.

- Lundberg, J. and Moberg, F. (2003). Mobile link organisms and ecosystem functioning: implications for ecosystem resilience and management. *Ecosystems* **6**: 0087-0098.
- MacKenzie, R. and Bruland, G. (2012). Nekton Communities in Hawaiian Coastal Wetlands: The Distribution and Abundance of Introduced Fish Species. *Estuaries and Coasts* **35**: 212-226.
- Manson, F. J., Loneragan, N. R., Harch, B. D., Skilleter, G. A. and Williams, L. (2005). A broadscale analysis of links between coastal fisheries production and mangrove extent: a case-study for northeastern Australia. *Fisheries Research* **74**: 69-85.
- McArdle, B. H. and Anderson, M. J. (2001). Fitting multivariate models to community data: a comment on distance based redundancy analysis. *Ecology* **82**: 290-297.
- Miranda, L. and Collazo, J. A. (1997). Food habits of 4 species of wading birds (Ardeidae) in a tropical mangrove swamp. *Colonial Waterbirds*: 413-418.
- Montgomery, S. S. (1990). Movements of juvenile eastern king prawns, Penaeus plebejus, and identification of stock along the east coast of Australia. *Fisheries Research* **9**: 189-208.
- Morton, R. M. (1990). Community structure, density and standing crop of fishes in a subtropical Australian mangrove area. *Marine Biology* **105**: 385-394.
- Ochwada-Doyle, F., Loneragan, N. R., Gray, C. A., Suthers, I. M. and Taylor, M. D. (2009). Complexity affects habitat preference and predation mortality in postlarval Penaeus plebejus: implications for stock enhancement. *Marine Ecology Progress Series* **380**: 161-171.
- Petchey, O. L., Downing, A. L., Mittelbach, G. G., Persson, L., Steiner, C. F., Warren, P. H. and Woodward, G. (2004). Species loss and the structure and functioning of multitrophic aquatic systems. *Oikos* **104**: 467-478.
- Pollard, D. A. and Hannan, J. C. (1994). The ecological effects of structural flood mitigation works on fish habitats and fish communities in the lower Clarence river system of southeastern Australia. *Estuaries* 17: 427-461.
- Pressey, R. L. and Middleton, M. J. (1982). Impacts of flood mitigation works on coastal wetlands in New South Wales. *Wetlands* **2**: 27-44.
- Pyke, G. H. and White, A. W. (2000). Factors influencing predation on eggs and tadpoles of the endangered Green and Golden Bell Frog *Litoria aurea* by the introduced Plague Minnow Gambusia holbrooki. *Australian Zoologist* **31**: 496-505.
- Racek, A. A. (1959). Prawn investigations in eastern Australia. *Research Bulletin of State Fisheries N.S.W.* **6**.
- Raposa, K. B. and Roman, C. T. (2003). Using gradients in tidal restriction to evaluate nekton community responses to salt marsh restoration. *Estuaries and Coasts* **26**: 198-205.
- Roman, C. T., Niering, W. A. and Warren, R. S. (1984). Salt marsh vegetation change in response to tidal restriction. *Environmental Management* **8**: 141-149.
- Ruello, N. V. (1975). Geographical distribution, growth and breeding migration of the eastern Australian king prawn *Penaeus plebejus* Hess. *Marine and Freshwater Research* **26**: 343-354.
- Sammut, J., Melville, M. D., Callinan, R. B. and Fraser, G. C. (1995). Estuarine acidification: impacts on aquatic biota of draining acid sulphate soils. *Australian Geographical Studies* 33: 89-100.
- Sammut, J., White, I. and Melville, D. (1996). Acidification of an estuarine tributary in eastern Australia due to drainage of acid sulphate soils. *Marine and Freshwater Research* **47**: 669-684.

- Sanaullah, M., Hjeltnes, B. and Ahmed, A. T. A. (2001). The relationship of some environmental factors and the Epizootic Ulceration Syndrome outbreaks in Beel Mahmoodpur, Faridpur, Bangladesh. *Asian Fisheries Science* **14**: 301-315.
- Skilleter, G. A., Olds, A., Loneragan, N. R. and Zharikov, Y. (2005). The value of patches of intertidal seagrass to prawns depends on their proximity to mangroves. *Marine Biology* 147: 353-365.
- Valentine-Rose, L., Cherry, J. A., Jacob Culp, J., Perez, K. E., Pollock, J. B., Arrington, D. A. and Layman, C. A. (2007). Floral and faunal differences between fragmented and unfragmented Bahamian tidal creeks. *Wetlands* **27**: 702-718.
- Williams, R. J. and Watford, F. A. (1997). Identification of structures restricting tidal flow in New South Wales, Australia. *Wetlands Ecology and Management* **5**: 87-97.
- Winning, G. and Saintilan, N. (2009). Vegetation changes in Hexham Swamp, Hunter River, New South Wales, since the construction of floodgates in 1971. *Cunninghamia* **11**: 185-194.
- Young, P. (1975). Preliminary observations on the environment and biology of juvenile king prawns (Penaeus plebejus) in Moreton Bay, Queensland. In: *Proceedings of the First Australian national prawn seminar. Australian Government Publishing Service, Canberra.* 18-36.
- Young, P. and Carpenter, S. (1977). Recruitment of postlarval penaeid prawns to nursery areas in Morton Bay, Queensland. *Marine and Freshwater Research* **28**: 745-773.
- Young, P. (1978). Moreton Bay, Queensland: a nursery area for juvenile penaeid prawns. Marine and Freshwater Research **29**: 55-75.

Appendices

Appendix 1 – Catch summary

			Grand		Ironbar	k Creek		Co	bbans Cre	eek	Mo	squito Cr	eek	Purg	gatory Cr	eek
Family	Scientific Name †	Common Name	Total	Total	Below	Lower	Upper	Total	Lower	Upper	Total	Lower	Upper	Total	Below	Lower
FISH																
Ambassidae	Ambassis iacksoniensis ^{E-M}	Port Jackson Glassfish	11.243	6,499	2.784	2.384	1.331	1.896	1.651	245	680	358	322	2.168	2.139	29
	Ambassis marianus E-M	Estuary perchlet	219	52	11	24	17	7	6	1	12	10	2	148	147	1
	Ambassis spp E-M	Glassfish	7,793	6.230	3.522	2.399	309	1.076	992	84	366	272	94	121	121	0
Anguillidae	Anguilla australis F-E*	Short-finned eel	15	3	0	-,	2	-,	0	0	0	0	0	12	0	12
. inguindae	Anguilla reinhardtii ^{F-E*}	Long-finned eel	7	1	0	0	- 1	ů	0	0	1	0	1	5	1	4
Clumaidaa	Hawklatsiahthus aastalugui E-M	Southarn Harring	10	1	0	0	0	1	1	0		0	0		0	-
Ciupeidae	Humandonhus vittatus EM*	Southern Herring	941	56	15	41	0	507	121	296	,	, ,	0	277	7	0
	nyperiopnus vinaius	Sandy Sprat	041	50	15	41	0	507	121	380	1	1	0	211	211	0
a	Potamaiosa richmonala	Freshwater nerring	3		0	0	0	0	0	0	0	0	0	3	3	0
Cyprinidae	Carassius auratus	Goldfish	92	22	0	0	22	0	0	0	0	0	0	70	9	61
	Cyprinus carpio	Common carp	28	0	0	0	0	0	0	0	0	0	0	28	1	27
Dasyatidae	Dasyatis spp ^{E-M}	Stingray	2	0	0	0	0	0	0	0	0	0	0	2	2	0
Engraulidae	Engraulis australis EM*	Australian anchovy	3	1	1	0	0	1	0	1	0	0	0	1	1	0
Galaxiidae	Galaxias maculatus F-E	Common jollytail	7	1	1	0	0	0	0	0	0	0	0	6	0	6
Gerreidae	Gerres subfasciatus E-M*	Silver biddy	447	285	188	78	19	61	25	36	14	12	2	87	52	35
Girellidae	Girella tricuspidata EM [*]	Luderick	23	0	0	0	0	1	1	0	12	6	6	10	10	0
Gobiidae	Acanthogobius flavimanus F-E	Yellowfin goby	128	38	26	12	0	5	4	1	56	44	12	29	29	0
	Afurcagobius tamarensis F-E	Tamar River Goby	6,496	3,073	1,074	1,943	56	765	595	170	990	691	299	1,668	1,667	1
	Arenigobius bifrenatus E-M	Bridled Goby	1.853	621	315	305	1	100	92	8	902	499	403	230	230	0
	Arenigobius frenatus E-M	Half-bridled goby	288	102	62	36	4	81	34	47	100	51	49	5	5	0
	Arenia obius snn E-M	Araniaabius undefined	200	202	02	20		0	0		100	0		0	0	0
	Compto controides a obiodas E-M	Oustar goby	86	20	14	15	0	21	27	4	12	7	6	12	12	0
	Environmental goologues	En minita and mhu	2 726	2.755	1.047	800	0	271	262	109	280	257	22	211	211	0
	Favonigobius exquisitus	Exquisite said goby	3,720	2,755	1,947	800	•	3/1	205	108	269	237	32	511	511	0
	Giossogobius biocellatus	Golden Goby	4	4	1	2	1	0	0	0	0	0	0	0	0	0
	Gobiomorphus australis	Striped gudgeon	2,274	1,469	47	251	1,171	2	2	0	4	3	1	799	111	688
	Gobiopterus semivestitus	Glass Goby	34,327	9,640	4,860	1,806	2,974	10,173	7,283	2,890	9,484	5,611	3,873	5,030	5,018	12
	Hypseleotris compressa ^{F-E}	Empire gudgeon	7,215	1,339	15	243	1,081	6	5	1	2	1	1	5,868	54	5,814
	Hypseleotris galii ^{F-E}	Firetailed Gudgeon	4	2	0	0	2	0	0	0	0	0	0	2	1	1
	Mugilogobius paludis ^{E-M}	Mangrove Goby	756	201	41	92	68	69	46	23	455	179	276	31	10	21
	Philypnodon grandiceps F-E	Flathead gudgeon	6,378	960	55	315	590	43	5	38	55	26	29	5,320	3,747	1,573
	Philypnodon macrostomus F-E	Dwarf flathead gudgeon	195	8	0	0	8	0	0	0	0	0	0	187	6	181
	Pseudogobius olorum F-E	Swan River goby	17,620	6,704	2,333	3,327	1,044	2,296	1,576	720	5,327	2,031	3,296	3,293	1,271	2,022
	Redigobius macrostoma E-M	Largemouth goby	13.249	4.053	824	2.429	800	1.515	555	960	4,506	2.250	2.256	3,175	2.832	343
	Taenioides purpurascens ^{EM}	Eel Goby	13	10	9	-,,	0	-,	0	0	3	1	2	0	0	0
Monodactvlidae	Monodactulus argenteus E-M	Silver batfish	0	1	1	0	0	ů	0	0	1	0	- 1	7	4	3
Mugilidaa	Ling amountog EM*	Elat tail mullat	4 409	1 770	1 169	508	04	1 070	811	257	1 060	560	401	400	40.4	5
wugnidae	Musil and also F-E*	Fiat-tail munici	4,400	2,575	1,100	100	1 006	1,079	10	251	1,000	205	491	477	424	05
	Mugu cepnaus	Sea Munet	3,632	2,575	1,289	190	1,090	15	10	3	343	203	118	921	820	95
Paralichthyidae	Pseudorhombus jenysu	Small-toothed flounder	21	8	2	6	0	8	7	1	0	0	0	5	5	0
Percichthyidae	Percalates colonorum	Estuary perch	1	0	0	0	0	0	0	0	0	0	0	1	0	1
	Percalates novemaculeata	Australian bass	21	10	0	5	5	0	0	0	7	6	1	4	4	0
Platycephalidae	Platycephalus fuscus E-M*	Dusky flathead	22	13	7	6	0	3	2	1	3	3	0	3	3	0
Poeciliidae	Gambusia holbrooki ^{F-E}	Mosquitofish	5,034	1,741	31	314	1,396	8	0	8	66	4	62	3,219	189	3,030
Pomatomidae	Pomatomus saltatrix E-M*	Tailor	26	3	2	0	1	6	2	4	1	0	1	16	16	0
Pseudomugilidae	Pseudomugil signifer F-E	Southern blue-eye	4,575	613	103	277	233	1,066	669	397	2,846	1,135	1,711	50	50	0
Scatophagidae	Scatophagus argus E-M	Spotted scat	2	0	0	0	0	0	0	0	1	0	1	1	1	0
Scorpaenidae	Centropogon australis E-M	Eastern fortescue	119	27	8	12	7	15	12	3	60	15	45	17	17	0
F	Notesthes robusta F-E	Bullrout	1		0	0	0	1	0	1	0	0	0	0	0	0
Sillaginidae	Sillano ciliata E-M*	Sand Whiting	66	17	4	0	4	11	8	3	22	17	5	16	16	0
Snagindae	A can then a a mus a ustralia E-M*	Valloufin broom	2 7 4 8	1 091	516	450	106	1 211	1 1 2 2		210	171	149	127	126	1
sparidae	Acaninopagrus australis	Tenowini bream	2,740	1,081	510	439	106	1,211	1,125	00	519	1/1	148	157	150	1
	Rhabdosargus sarba	Tarwhine	172	80	9	67	4	14	14	0	53	17	36	25	25	0
Terapontidae	Pelates sexlineatus	Eastern striped grunter	1	0	0	0	0	0	0	0	0	0	0	1	0	1
	Terapon jarbua	Grunter	5	3	2	1	0	2	1	1	0	0	0	0	0	0
Tetraodontidae	Tetractenos glaber	Smooth toadfish	11	7	7	0	0	4	4	0	0	0	0	0	0	0
	Tetractenos hamiltoni ^{E-M}	Common toadfish	2	0	0	0	0	2	2	0	0	0	0	0	0	0
	ΤΟ ΤΑΙ	ABUNDANCE OF FISH	136 432	52 109	21 294	18 360	12 455	22 450	15 960	6 490	28 043	14 461	13 582	33 830	19 863	13 967
	TOTAL NU	MREP OF FISH SPECIES	130,432	43	36	34	31	22,430	33	30	20,045	31	32	35,050	42	25
	IO IAL NO		55	45	50	54	51	50	55	50	55	51	52	40		20
DECLOOD CRUS																
DECAPODCRUS	IACEA EM															
Alpheidae	Alpheus spp	Pistol Shrimp	1,586	672	283	381	8	264	189	75	502	335	167	148	148	0
Grapsidae	Grapsidae Spp	Marsh crab	102	40	22	18	0	8	5	3	26	11	15	28	20	8
	Parasesarma erythrodactyia	Grapsid crab	46	22	13	5	4	6	5	1	11	7	4	7	7	0
Hymenossomatidae	Amarinus lacustris F-E	False Spider Crab	317	30	15	14	1	21	20	1	31	21	10	235	16	219
Ocypodidae	Ocypode Spp E-M	Ghost Crab	14	4	4	0	0	6	5	1	1	0	1	3	3	0
	Ocypode Spp ^{E-M}	Ghost Crab	1	0	0	0	0	0	0	0	0	0	0	1	1	0
Palaemonidae	Macrobrachium intermedium F-E	Striped shrimp	47	9	1	2	6	14	0	14	12	9	3	12	12	0
	Macrobrachium novaehollandiae	Long-armed prawn	1	0	0	0	0	0	0	0	1	1	0	0	0	0
	Palaemon debilis E-M	Grass shrimp	39.796	14,100	6,184	5,579	2.337	14,783	7,817	6,966	7,685	4,171	3.514	3,228	3,154	74
	Palaemonidae spp E-M	Shrimp	15 103	7,018	3,756	3,253	0	1.430	531	890	6,137	2.088	4.049	518	430	79
Panaaidae	Malicartus nlabajus EM*	Eastern king proup	1 225	.,010	100	240	27	292	101	101	29127	110	162	104	104	
1 chacidae	Matananaana haw EM*	Graagybaak receive	1,335	200	140	349	5/	122	191	191	401	116	105	257	217	40
	Meiapenaeus bennemae	Greasyback prawn	2,038	010	146	453	11	132	93	39	939	458	481	357	31/	40
	meiapenaeus macleayi	school prawn	32,082	14,709	2,650	9,136	2,923	3,130	1,459	1,671	8,250	3,862	4,388	5,993	5,654	339
Sergestidae	Acetes sibogae australis 🗠	Pink Shrimp	142,613	56,165	39,048	8,346	8,771	19,660	17,254	2,406	25,607	18,561	7,046	41,181	41,047	134
	TO TAL ABUN	DANCE OF DECAPODS	235.081	93,945	52,302	27,536	14,107	39.836	27,569	12,267	49,483	29,642	19,841	51,817	50,924	893
	TO TAL NUMBER	OF DECAPOD SPECIES	14	12	12	11	10	12	11	12	13	12	12	13	13	7

E-M = estuarine-marine dwelling; F-E = occupies fresh and saltwater; F = purely freshwater dwelling. * = commercially important species.

Appendix 2 – Prawn CPUE plots

Figure A2.1 Change in mean (±S.E.) catch per unit effort (CPUE) of juvenile Eastern king prawn (*Melicertus plebejus*) in lower Hunter River study sites







Figure A2.2 Change in mean (±S.E.) catch per unit effort (CPUE) of juvenile School prawn (*Metapenaeus macleayi*) in lower Hunter River study sites



Other titles in this series

- No. 1 Andrew, N.L., Graham, K.J., Hodgson, K.E. and Gordon, G.N.G., 1998. Changes after 20 years in relative abundance and size composition of commercial fishes caught during fishery independent surveys on SEF trawl grounds. Final Report to Fisheries Research and Development Corporation. Project no. 96/139.
- No. 2 Virgona, J.L., Deguara, K.L., Sullings, D.J., Halliday, I. and Kelly, K., 1998. Assessment of the stocks of sea mullet in New South Wales and Queensland waters.
- No. 3 Stewart, J., Ferrell, D.J. and Andrew, N.L., 1998. Ageing Yellowtail (*Trachurus novaezelandiae*) and Blue Mackerel (*Scomber australasicus*) in New South Wales.
- No. 4 Pethebridge, R., Lugg, A. and Harris, J., 1998. Obstructions to fish passage in New South Wales South Coast streams. 70pp.
- No. 5 Kennelly, S.J. and Broadhurst, M.K., 1998. Development of by-catch reducing prawn-trawls and fishing practices in NSW's prawn-trawl fisheries (and incorporating an assessment of the effect of increasing mesh size in fish trawl gear). 18pp + appendices.
- No. 6 Allan, G.L. and Rowland, S.J., 1998. Fish meal replacement in aquaculture feeds for silver perch. 237pp + appendices.
- No. 7 Allan, G.L., 1998. Fish meal replacement in aquaculture feeds: subprogram administration. 54pp + appendices.
- No. 8 Heasman, M.P., O'Connor, W.A. and O'Connor, S.J., 1998. Enhancement and farming of scallops in NSW using hatchery produced seedstock. 146pp.
- No. 9 Nell, J.A., McMahon, G.A. and Hand, R.E., 1998. Tetraploidy induction in Sydney rock oysters. 25pp.
- No. 10 Nell, J.A. and Maguire, G.B., 1998. Commercialisation of triploid Sydney rock and Pacific oysters. Part 1: Sydney rock oysters. 122pp.
- No. 11 Watford, F.A. and Williams, R.J., 1998. Inventory of estuarine vegetation in Botany Bay, with special reference to changes in the distribution of seagrass. 51pp.
- No. 12 Andrew, N.L., Worthington D.G., Brett, P.A. and Bentley N., 1998. Interactions between the abalone fishery and sea urchins in New South Wales.
- No. 13 Jackson, K.L. and Ogburn, D.M., 1999. Review of depuration and its role in shellfish quality assurance. 77pp.
- No. 14 Fielder, D.S., Bardsley, W.J. and Allan, G.L., 1999. Enhancement of Mulloway (*Argyrosomus japonicus*) in intermittently opening lagoons. 50pp + appendices.
- No. 15 Otway, N.M. and Macbeth, W.G., 1999. The physical effects of hauling on seagrass beds. 86pp.
- No. 16 Gibbs, P., McVea, T. and Louden, B., 1999. Utilisation of restored wetlands by fish and invertebrates. 142pp.
- No. 17 Ogburn, D. and Ruello, N., 1999. Waterproof labelling and identification systems suitable for shellfish and other seafood and aquaculture products. Whose oyster is that? 50pp.
- No. 18 Gray, C.A., Pease, B.C., Stringfellow, S.L., Raines, L.P. and Walford, T.R., 2000. Sampling estuarine fish species for stock assessment. Includes appendices by D.J. Ferrell, B.C. Pease, T.R. Walford, G.N.G. Gordon, C.A. Gray and G.W. Liggins. 194pp.
- No. 19 Otway, N.M. and Parker, P.C., 2000. The biology, ecology, distribution, abundance and identification of marine protected areas for the conservation of threatened Grey Nurse Sharks in south east Australian waters. 101pp.
- No. 20 Allan, G.L. and Rowland, S.J., 2000. Consumer sensory evaluation of silver perch cultured in ponds on meat meal based diets. 21pp + appendices.
- No. 21 Kennelly, S.J. and Scandol, J. P., 2000. Relative abundances of spanner crabs and the development of a population model for managing the NSW spanner crab fishery. 43pp + appendices.
- No. 22 Williams, R.J., Watford, F.A. and Balashov, V., 2000. Kooragang Wetland Rehabilitation Project: History of changes to estuarine wetlands of the lower Hunter River. 82pp.
- No. 23 Survey Development Working Group, 2000. Development of the National Recreational and Indigenous Fishing Survey. Final Report to Fisheries Research and Development Corporation. (Volume 1 36pp + Volume 2 attachments).
- No.24 Rowling, K.R and Raines, L.P., 2000. Description of the biology and an assessment of the fishery of Silver Trevally *Pseudocaranx dentex* off New South Wales. 69pp.
- No. 25 Allan, G.L., Jantrarotai, W., Rowland, S., Kosuturak, P. and Booth, M., 2000. Replacing fishmeal in aquaculture diets. 13pp.
- No. 26 Gehrke, P.C., Gilligan, D.M. and Barwick, M., 2001. Fish communities and migration in the Shoalhaven River Before construction of a fishway. 126pp.

- Boys
- No. 27 Rowling, K.R. and Makin, D.L., 2001. Monitoring of the fishery for Gemfish *Rexea solandri*, 1996 to 2000. 44pp.
- No. 28 Otway, N.M., 1999. Identification of candidate sites for declaration of aquatic reserves for the conservation of rocky intertidal communities in the Hawkesbury Shelf and Batemans Shelf Bioregions. 88pp.
- No. 29 Heasman, M.P., Goard, L., Diemar, J. and Callinan, R., 2000. Improved Early Survival of Molluscs: Sydney Rock Oyster (*Saccostrea glomerata*). 63pp.
- No. 30 Allan, G.L., Dignam, A and Fielder, S., 2001. Developing Commercial Inland Saline Aquaculture in Australia: Part 1. R&D Plan.
- No. 31 Allan, G.L., Banens, B. and Fielder, S., 2001. Developing Commercial Inland Saline Aquaculture in Australia: Part 2. Resource Inventory and Assessment. 33pp.
- No. 32 Bruce, A., Growns, I. and Gehrke, P., 2001. Woronora River Macquarie Perch Survey. 116pp.
- No. 33 Morris, S.A., Pollard, D.A., Gehrke, P.C. and Pogonoski, J.J., 2001. Threatened and Potentially Threatened Freshwater Fishes of Coastal New South Wales and the Murray-Darling Basin. 177pp.
- No. 34 Heasman, M.P., Sushames, T.M., Diemar, J.A., O'Connor, W.A. and Foulkes, L.A., 2001. Production of Micro-algal Concentrates for Aquaculture Part 2: Development and Evaluation of Harvesting, Preservation, Storage and Feeding Technology. 150pp + appendices.
- No. 35 Stewart, J. and Ferrell, D.J., 2001. Mesh selectivity in the NSW demersal trap fishery. 86pp.
- No. 36 Stewart, J., Ferrell, D.J., van der Walt, B., Johnson, D. and Lowry, M., 2001. Assessment of length and age composition of commercial kingfish landings. 49pp.
- No. 37 Gray, C.A. and Kennelly, S.J., 2001. Development of discard-reducing gears and practices in the estuarine prawn and fish haul fisheries of NSW. 151pp.
- No. 38 Murphy, J.J., Lowry, M.B., Henry, G.W. and Chapman, D., 2002. The Gamefish Tournament Monitoring Program 1993 to 2000. 93pp.
- No. 39 Kennelly, S.J. and McVea, T.A. (Ed), 2002. Scientific reports on the recovery of the Richmond and Macleay Rivers following fish kills in February and March 2001. 325pp.
- No. 40 Pollard, D.A. and Pethebridge, R.L., 2002. Report on Port of Botany Bay Introduced Marine Pest Species Survey. 69pp.
- No. 41 Pollard, D.A. and Pethebridge, R.L., 2002. Report on Port Kembla Introduced Marine Pest Species Survey. 72pp.
- No. 42 O'Connor, W.A, Lawler, N.F. and Heasman, M.P., 2003. Trial farming the akoya pearl oyster, *Pinctada imbricata*, in Port Stephens, NSW. 170pp.
- No. 43 Fielder, D.S. and Allan, G.L., 2003. Improving fingerling production and evaluating inland saline water culture of snapper, *Pagrus auratus*. 62pp.
- No. 44 Astles, K.L., Winstanley, R.K., Harris, J.H. and Gehrke, P.C., 2003. Experimental study of the effects of cold water pollution on native fish. 55pp.
- No. 45 Gilligan, D.M., Harris, J.H. and Mallen-Cooper, M., 2003. Monitoring changes in the Crawford River fish community following replacement of an effective fishway with a vertical-slot fishway design: Results of an eight year monitoring program. 80pp.
- No. 46 Pollard, D.A. and Rankin, B.K., 2003. Port of Eden Introduced Marine Pest Species Survey. 67pp.
- No. 47 Otway, N.M., Burke, A.L., Morrison, NS. and Parker, P.C., 2003. Monitoring and identification of NSW Critical Habitat Sites for conservation of Grey Nurse Sharks. 62pp.
- No. 48 Henry, G.W. and Lyle, J.M. (Ed), 2003. The National Recreational and Indigenous Fishing Survey. 188 pp.
- No. 49 Nell, J.A., 2003. Selective breeding for disease resistance and fast growth in Sydney rock oysters. 44pp. (Also available – a CD-Rom published in March 2004 containing a collection of selected manuscripts published over the last decade in peer-reviewed journals).
- No. 50 Gilligan, D. and Schiller, S., 2003. Downstream transport of larval and juvenile fish. 66pp.
- No. 51 Liggins, G.W., Scandol, J.P. and Kennelly, S.J., 2003. Recruitment of Population Dynamacist. 44pp.
- No. 52 Steffe, A.S. and Chapman, J.P., 2003. A survey of daytime recreational fishing during the annual period, March 1999 to February 2000, in Lake Macquarie, New South Wales. 124pp.
- No. 53 Barker, D. and Otway, N., 2003. Environmental assessment of zinc coated wire mesh sea cages in Botany Bay NSW. 36pp.
- No. 54 Growns, I., Astles, A. and Gehrke, P., 2003. Spatial and temporal variation in composition of riverine fish communities. 24pp.
- No. 55 Gray, C. A., Johnson, D.D., Young, D.J. and Broadhurst, M. K., 2003. Bycatch assessment of the Estuarine Commercial Gill Net Fishery in NSW. 58pp.
- No. 56 Worthington, D.G. and Blount, C., 2003. Research to develop and manage the sea urchin fisheries of NSW and eastern Victoria. 182pp.

- No. 57 Baumgartner, L.J., 2003. Fish passage through a Deelder lock on the Murrumbidgee River, Australia. 34pp.
- No. 58 Allan, G.L., Booth, M.A., David A.J. Stone, D.A.J. and Anderson, A.J., 2004. Aquaculture Diet Development Subprogram: Ingredient Evaluation. 171pp.
- No. 59 Smith, D.M., Allan, G.L. and Booth, M.A., 2004. Aquaculture Diet Development Subprogram: Nutrient Requirements of Aquaculture Species. 220pp.
- No. 60 Barlow, C.G., Allan, G.L., Williams, K.C., Rowland, S.J. and Smith, D.M., 2004. Aquaculture Diet Development Subprogram: Diet Validation and Feeding Strategies. 197pp.
- No. 61 Heasman, M.H., 2004. Sydney Rock Oyster Hatchery Workshop 8 9 August 2002, Port Stephens, NSW. 115pp.
- No. 62 Heasman, M., Chick, R., Savva, N., Worthington, D., Brand, C., Gibson, P. and Diemar, J., 2004. Enhancement of populations of abalone in NSW using hatchery-produced seed. 269pp.
- No. 63 Otway, N.M. and Burke, A.L., 2004. Mark-recapture population estimate and movements of Grey Nurse Sharks. 53pp.
- No. 64 Creese, R.G., Davis, A.R. and Glasby, T.M., 2004. Eradicating and preventing the spread of the invasive alga *Caulerpa taxifolia* in NSW. 110pp.
- No. 65 Baumgartner, L.J., 2004. The effects of Balranald Weir on spatial and temporal distributions of lower Murrumbidgee River fish assemblages. 30pp.
- No. 66 Heasman, M., Diggles, B.K., Hurwood, D., Mather, P., Pirozzi, I. and Dworjanyn, S., 2004. Paving the way for continued rapid development of the flat (angasi) oyster (*Ostrea angasi*) farming in New South Wales. 40pp.

ISSN 1449-9967 (NSW Department of Primary Industries – Fisheries Final Report Series)

- No. 67 Kroon, F.J., Bruce, A.M., Housefield, G.P. and Creese, R.G., 2004. Coastal floodplain management in eastern Australia: barriers to fish and invertebrate recruitment in acid sulphate soil catchments. 212pp.
- No. 68 Walsh, S., Copeland, C. and Westlake, M., 2004. Major fish kills in the northern rivers of NSW in 2001: Causes, Impacts & Responses. 55pp.
- No. 69 Pease, B.C. (Ed), 2004. Description of the biology and an assessment of the fishery for adult longfinned eels in NSW. 168pp.
- No. 70 West, G., Williams, R.J. and Laird, R., 2004. Distribution of estuarine vegetation in the Parramatta River and Sydney Harbour, 2000. 37pp.
- No. 71 Broadhurst, M.K., Macbeth, W.G. and Wooden, M.E.L., 2005. Reducing the discarding of small prawns in NSW's commercial and recreational prawn fisheries. 202pp.
- No. 72. Graham, K.J., Lowry, M.B. and Walford, T.R., 2005. Carp in NSW: Assessment of distribution, fishery and fishing methods. 88pp.
- No. 73 Stewart, J., Hughes, J.M., Gray, C.A. and Walsh, C., 2005. Life history, reproductive biology, habitat use and fishery status of eastern sea garfish (*Hyporhamphus australis*) and river garfish (*H. regularis ardelio*) in NSW waters. 180pp.
- No. 74 Growns, I. and Gehrke, P., 2005. Integrated Monitoring of Environmental Flows: Assessment of predictive modelling for river flows and fish. 33pp.
- No. 75 Gilligan, D., 2005. Fish communities of the Murrumbidgee catchment: Status and trends. 138pp.
- No. 76 Ferrell, D.J., 2005. Biological information for appropriate management of endemic fish species at Lord Howe Island. 18 pp.
- No. 77 Gilligan, D., Gehrke, P. and Schiller, C., 2005. Testing methods and ecological consequences of largescale removal of common carp. 46pp.
- No. 78 Boys, C.A., Esslemont, G. and Thoms, M.C., 2005. Fish habitat and protection in the Barwon-Darling and Paroo Rivers. 118pp.
- No. 79 Steffe, A.S., Murphy, J.J., Chapman, D.J. and Gray, C.C., 2005. An assessment of changes in the daytime recreational fishery of Lake Macquarie following the establishment of a 'Recreational Fishing Haven'. 103pp.
- No. 80 Gannassin, C. and Gibbs, P., 2005. Broad-Scale Interactions Between Fishing and Mammals, Reptiles and Birds in NSW Marine Waters. 171pp.
- No. 81 Steffe, A.S., Murphy, J.J., Chapman, D.J., Barrett, G.P. and Gray, C.A., 2005. An assessment of changes in the daytime, boat-based, recreational fishery of the Tuross Lake estuary following the establishment of a 'Recreational Fishing Haven'. 70pp.
- No. 82 Silberschnieder, V. and Gray, C.A., 2005. Arresting the decline of the commercial and recreational fisheries for mulloway (*Argyrosomus japonicus*). 71pp.
- No. 83 Gilligan, D., 2005. Fish communities of the Lower Murray-Darling catchment: Status and trends. 106pp.

- No. 84 Baumgartner, L.J., Reynoldson, N., Cameron, L. and Stanger, J., 2006. Assessment of a Dual-frequency Identification Sonar (DIDSON) for application in fish migration studies. 33pp.
- No. 85 Park, T., 2006. FishCare Volunteer Program Angling Survey: Summary of data collected and recommendations. 41pp.
- No. 86 Baumgartner, T., 2006. A preliminary assessment of fish passage through a Denil fishway on the Edward River, Australia. 23pp.
- No. 87 Stewart, J., 2007. Observer study in the Estuary General sea garfish haul net fishery in NSW. 23pp.
- No. 88 Faragher, R.A., Pogonoski, J.J., Cameron, L., Baumgartner, L. and van der Walt, B., 2007. Assessment of a stocking program: Findings and recommendations for the Snowy Lakes Trout Strategy. 46pp.
- No. 89 Gilligan, D., Rolls, R., Merrick, J., Lintermans, M., Duncan, P. and Kohen, J., 2007. Scoping knowledge requirements for Murray crayfish (*Euastacus armatus*). Final report to the Murray Darling Basin Commission for Project No. 05/1066 NSW 103pp.
- No. 90 Kelleway, J., Williams. R.J. and Allen, C.B., 2007. An assessment of the saltmarsh of the Parramatta River and Sydney Harbour. 100pp.
- No. 91 Williams, R.J. and Thiebaud, I., 2007. An analysis of changes to aquatic habitats and adjacent land-use in the downstream portion of the Hawkesbury Nepean River over the past sixty years. 97pp.
- No. 92 Baumgartner, L., Reynoldson, N., Cameron, L. and Stanger, J. The effects of selected irrigation practices on fish of the Murray-Darling Basin. 90pp.
- No. 93 Rowland, S.J., Landos, M., Callinan, R.B., Allan, G.L., Read, P., Mifsud, C., Nixon, M., Boyd, P. and Tally, P., 2007. Development of a health management strategy for the Silver Perch Aquaculture Industry. 219pp.
- No. 94 Park, T., 2007. NSW Gamefish Tournament Monitoring Angling Research Monitoring Program. Final report to the NSW Recreational Fishing Trust. 142pp.
- No. 95 Heasman, M.P., Liu, W., Goodsell, P.J., Hurwood D.A. and Allan, G.L., 2007. Development and delivery of technology for production, enhancement and aquaculture of blacklip abalone (*Haliotis rubra*) in New South Wales. 226pp.
- No. 96 Ganassin, C. and Gibbs, P.J., 2007. A review of seagrass planting as a means of habitat compensation following loss of seagrass meadow. 41pp.
- No. 97 Stewart, J. and Hughes, J., 2008. Determining appropriate harvest size at harvest for species shared by the commercial trap and recreational fisheries in New South Wales. 282pp.
- No. 98 West, G. and Williams, R.J., 2008. A preliminary assessment of the historical, current and future cover of seagrass in the estuary of the Parramatta River. 61pp.
- No. 99 Williams, D.L. and Scandol, J.P., 2008. Review of NSW recreational fishing tournament-based monitoring methods and datasets. 83pp.
- No. 100 Allan, G.L., Heasman, H. and Bennison, S., 2008. Development of industrial-scale inland saline aquaculture: Coordination and communication of R&D in Australia. 245pp.
- No. 101 Gray, C.A and Barnes, L.M., 2008. Reproduction and growth of dusky flathead (*Platycephalus fuscus*) in NSW estuaries. 26pp.
- No. 102 Graham, K.J., 2008. The Sydney inshore trawl-whiting fishery: codend selectivity and fishery characteristics. 153pp.
- No. 103 Macbeth, W.G., Johnson, D.D. and Gray, C.A., 2008. Assessment of a 35-mm square-mesh codend and composite square-mesh panel configuration in the ocean prawn-trawl fishery of northern New South Wales. 104pp.
- No. 104 O'Connor, W.A., Dove, M. and Finn, B., 2008. Sydney rock oysters: Overcoming constraints to commercial scale hatchery and nursery production. 119pp.
- No. 105 Glasby, T.M. and Lobb, K., 2008. Assessing the likelihoods of marine pest introductions in Sydney estuaries: A transport vector approach. 84pp.
- No. 106 Rotherham, D., Gray, C.A., Underwood, A.J., Chapman, M.G. and Johnson, D.D., 2008. Developing fishery-independent surveys for the adaptive management of NSW's estuarine fisheries. 135pp.
- No. 107 Broadhurst, M., 2008. Maximising the survival of bycatch discarded from commercial estuarine fishing gears in NSW. 192pp.
- No. 108 Gilligan, D., McLean, A. and Lugg, A., 2009. Murray Wetlands and Water Recovery Initiatives: Rapid assessment of fisheries values of wetlands prioritised for water recovery. 69pp.
- No. 109 Williams, R.J. and Thiebaud, I., 2009. Occurrence of freshwater macrophytes in the catchments of the Parramatta River, Lane Cove River and Middle Harbour Creek, 2007 2008. 75pp.
- No. 110 Gilligan, D., Vey, A. and Asmus, M., 2009. Identifying drought refuges in the Wakool system and assessing status of fish populations and water quality before, during and after the provision of environmental, stock and domestic flows. 56pp.

ISSN 1837-2112 (Industry & Investment NSW – Fisheries Final Report Series)

- No. 111 Gray, C.A., Scandol. J.P., Steffe, A.S. and Ferrell, D.J., 2009. Australian Society for Fish Biology Annual Conference & Workshop 2008: Assessing Recreational Fisheries; Current and Future Challenges. 54pp.
- No. 112 Otway, N.M. Storrie, M.T., Louden, B.M. and Gilligan, J.J., 2009. Documentation of depth-related migratory movements, localised movements at critical habitat sites and the effects of scuba diving for the east coast grey nurse shark population. 90pp.
- No. 113 Creese, R.G., Glasby, T.M., West, G. and Gallen, C., 2009. Mapping the habitats of NSW estuaries. 95pp.
- No. 114 Macbeth, W.G., Geraghty, P.T., Peddemors, V.M. and Gray, C.A., 2009. Observer-based study of targeted commercial fishing for large shark species in waters off northern New South Wales. 82pp.
- No. 115 Scandol, J.P., Ives, M.C. and Lockett, M.M., 2009. Development of national guidelines to improve the application of risk-based methods in the scope, implementation and interpretation of stock assessments for data-poor species. 186pp.
- No. 116 Baumgartner, L., Bettanin, M., McPherson, J., Jones, M., Zampatti, B. and Kathleen Beyer., 2009. Assessment of an infrared fish counter (Vaki Riverwatcher) to quantify fish migrations in the Murray-Darling Basin. 47pp.
- No. 117 Astles, K., West, G., and Creese, R.G., 2010. Estuarine habitat mapping and geomorphic characterisation of the Lower Hawkesbury river and Pittwater estuaries. 229pp.
- No. 118 Gilligan, D., Jess, L., McLean, G., Asmus, M., Wooden, I., Hartwell, D., McGregor, C., Stuart, I., Vey, A., Jefferies, M., Lewis, B. and Bell, K., 2010. Identifying and implementing targeted carp control options for the Lower Lachlan Catchment. 126pp.
- No. 119 Montgomery, S.S., Walsh, C.T., Kesby, C.L and Johnson, D.D., 2010. Studies on the growth and mortality of school prawns. 90pp.
- No. 120 Liggins, G.W. and Upston, J., 2010. Investigating and managing the *Perkinsus*-related mortality of blacklip abalone in NSW. 182pp.
- No. 121 Knight, J., 2010. The feasibility of excluding alien redfin perch from Macquarie perch habitat in the Hawkesbury-Nepean Catchment. 53pp.
- No. 122 Ghosn, D., Steffe, A., Murphy, J., 2010. An assessment of the effort and catch of shore and boat-based recreational fishers in the Sydney Harbour estuary over the 2007/08 summer period. 60pp.
- No. 123 Rourke, M. and Gilligan, D., 2010. Population genetic structure of freshwater catfish (*Tandanus tandanus*) in the Murray-Darling Basin and coastal catchments of New South Wales: Implications for future re-stocking programs. 74pp.
- No. 124 Tynan, R., Bunter, K. and O'Connor, W., 2010. Industry Management and Commercialisation of the Sydney Rock Oyster Breeding Program. 21pp.
- No. 125 Lowry, M., Folpp, H., Gregson, M. and McKenzie, R., 2010. Assessment of artificial reefs in Lake Macquarie NSW. 47pp.
- No. 126 Howell, T. and Creese, R., 2010. Freshwater fish communities of the Hunter, Manning, Karuah and Macquarie-Tuggerah catchments: a 2004 status report. 93pp.
- No. 127 Gilligan, D., Rodgers, M., McGarry, T., Asmus, M. and Pearce, L., 2010. The distribution and abundance of two endangered fish species in the NSW Upper Murray Catchment. 34pp.
- No. 128 Gilligan, D., McGarry, T. and Carter, S., 2010. A scientific approach to developing habitat rehabilitation strategies in aquatic environments: A case study on the endangered Macquarie perch (*Macquaria australasica*) in the Lachlan catchment. 61pp.
- No. 129 Stewart, J., Hughes, J., McAllister, J., Lyle, J. and MacDonald, M., 2011. Australian salmon (*Arripis trutta*): Population structure, reproduction, diet and composition of commercial and recreational catches. 257 pp.

ISSN 1837-2112 (Fisheries Final Report Series)

- No. 130 Boys, C., Glasby, T., Kroon, F., Baumgartner, L., Wilkinson, K., Reilly, G. and Fowler, T., 2011. Case studies in restoring connectivity of coastal aquatic habitats: floodgates, box culvert and rock-ramp fishway. 75pp.
- No. 131 Steffe, A.S. and Murphy, J.J., 2011. Recreational fishing surveys in the Greater Sydney Region. 122pp.
- No. 132 Robbins, W.D., Peddemors, V.M. and Kennelly, S.J., 2012. Assessment of shark sighting rates by aerial beach patrols. 38pp.
- No. 133 Boys, C.A. and Williams, R.J., 2012. Fish and decapod assemblages in Kooragang Wetlands: the impact of tidal restriction and responses to culvert removal. 80pp.
- No. 134 Boys, C.A, Baumgartner,L., Rampano, B., Alexander, T., Reilly, G., Roswell, M., Fowler, T and Lowry. M. 2012. Development of fish screening criteria for water diversions in the Murray-Darling Basin. 62pp.

- No. 135 Boys, C.A, Southwell, M., Thoms, M., Fowler, T, Thiebaud, I., Alexander, T. and Reilly, G. 2012. Evaluation of aquatic rehabilitation in the Bourke to Brewarrina Demonstration Reach, Barwon-Darling River, Australia. 133pp
- No. 136 Baumgartner, L., McPherson, B., Doyle, J., Cory, J., Cinotti, N. and Hutchison, J. 2013. Quantifying and mitigating the impacts of weirs on downstream passage of native fish in the Murray-Darling Basin. 79pp.
- No. 137 Boys, C.A, Baumgartner, B., Miller, B., Deng, Z., Brown, R. and Pflugrath, B. 2013. Protecting downstream migrating fish at mini hydropower and other river infrastructure. 93pp.
- No. 138 Hughes, J.M. and Stewart, J. 2013. Assessment of barotrauma and its mitigation measures on the behaviour and survival of snapper and mulloway. 152pp.
- No. 139 Ochwada-Doyle, F.A., McLeod, J., Barrett, G., Clarke, G. and Gray, C.A., 2014. Assessment of recreational fishing in three recreational fishing havens in New South Wales. 29pp.
- No. 140 Walsh, C. T., Rodgers, M. P., Thorne, N. J. and Robinson, W. A., 2013. Thermoshock Fish Mortality Investigation. 32pp.
- No. 141 Boys, C.A., Navarro, A., Robinson, W., Fowler, T., Chilcott, S., Miller, B., Pflugrath, B., Baumgartner, L.J., McPherson, J., Brown, R. and Deng, Z., 2014. Downstream fish passage criteria for hydropower and irrigation infrastructure in the Murray-Darling Basin. 119pp.
- No. 142 Cameron, L., Baumgartner, L. and Miners, B., 2012. Assessment of Australian bass restocking in the upper Snowy River. 102pp.
- No. 143 Walsh, C., Rodgers, M., Robinson, W. and Gilligan, D., 2014. Evaluation of the effectiveness of the Tallowa Dam Fishway. 89pp.

ISSN 2204-8669 (NSW Trade & Investment – Fisheries Final Report Series)

- No. 144 Ghosn, D.L., Collins, D.P. and Gould, A., 2015. The NSW Game Fish Tournament Monitoring Program 1994 to 2013: A summary of data and assessment of the role and design. 200pp.
- No. 145 Boys, C., 2015. Changes in fish and crustacean assemblages in tidal creeks of Hexham Swamp following the staged opening of Ironbark Creek floodgates. 47pp.